ECONOMIC PAPERS

COMMISSION OF THE EUROPEAN COMMUNITIES • DIRECTORATE-GENERAL FOR ECONOMIC AND FINANCIAL AFFAIRS

Number 96

October 1992

Budgeting Procedures and Fiscal Performance in the European Communities

Jürgen von Hagen*



"Economic Papers" are written by the Staff of the Directorate-General for Economic and Financial Affairs, or by experts working in association with them. The "Papers" are intended to increase awareness of the technical work being done by the staff and to seek comments and suggestions for further analyses. They may not be quoted without authorisation. Views expressed represent exclusively the positions of the author and do not necessarily correspond with those of the Commission of the European Communities. Comments and enquiries should be addressed to:

The Directorate-General for Economic and Financial Affairs, Commission of the European Communities, 200, rue de la Loi 1049 Brussels, Belgium

ECONOMIC PAPERS

Number 96

October 1992

Budgeting Procedures and Fiscal Performance in the European Communities

Jürgen von Hagen*

* University of Mannheim, Indiana University School of Business, and CEPR

This research was supported by the Commission of European Communities. The views expressed in this paper are those of the author and do not reflect views of the Commission of the European Communities. Helpful comments by Michele Fratianni and Suzanne Lohmann are gratefully acknowledged.

Abstract

We investigate the role of budgeting procedures for fiscal performance. Using 1970s and 1980s EC fiscal data and expert characterizations of budgeting procedures, we find strong empirical support for the 'structural hypothesis' that a budgeting process with strategic dominance of the prime or finance (or treasury) minister over the spending ministers, limits on parliamentary amendment power, and limiting changes during the execution process is strongly conducive to fiscal discipline. In contrast, the role of long-term fiscal constraints in achieving fiscal discipline, while generally positive, is not statistically significant. The results suggest that institutional reform of the budgeting process is a promising avenue to achieve and maintain a larger degree of fiscal discipline.

Non-technical Abstract

The approach of a monetary union in Europe has raised concerns about the appropriate fiscal policy regime in the EC. One important worry is that a systematic lack of fiscal stability of some members of the European monetary union might create political pressures for monetary expansions which the European central bank will find hard to escape, resulting in persistent inflation. As a result, the revisions of the Treaties of Rome adopted in Maastricht in December 1991 incorporate a procedure to supervise the fiscal performance of the members by the Community and to increase fiscal discipline.

The basic approach taken in the Maastricht Accord is one of controlling, ex-ante and ex-post, the Member States' fiscal performance by the Community, spelling out budget criteria and procedures for monitoring fiscal performance in the EC. To be effective, such a strategy must rely on the credibility of the penalties it entails for violating these criteria. Experience with budget norms in the U.S. suggests that governments find ways to circumvent fiscal restraints in practice, with the result that they are largely ineffective. In the European context, the absence of a strict enforcement mechanism of fiscal constraints among sovereign nations other than the threat or possible application of peer pressure and financial sanctions raises the problem of how to ensure the prospects of budget criteria to be successful.

This study looks at the issue from another perspective. We start from the presumption that budgeting procedures, i.e., the rules according to which budgets are drafted by a government, amended and passed by parliament, and implemented by the government, are important for the degree of fiscal stability attained. In other words, we claim that institutions shape the outcome of the political processes evolving within them. One variant of this claim - that the greater credibility of an independent central bank's commitment to price stability leads to lower inflation rates - is, of course, one of the important justifications for the European monetary union itself. Our main proposition is that a budgeting procedure enabling a government to commit itself to fiscal discipline is an essential condition for fiscal stability. Commitment mechanisms are important on all three levels of the budget process, the bargaining within the cabinet of ministers, the passing of the budget law through parliament, and the execution of the budget.

Budgeting in government is a process of allocating resources to specific political programs and distributing the cost over current and future tax payers. There are (at least) two aspects of this process which generate problems of fiscal discipline and give rise to the importance of institutions. One is the difference between the short-run and the long-run net benefits of fiscal programs, which, if policy makers discount the future, induces a bias towards shifting tax burdens to future tax payers via deficit financing. The other is the difference between the perceived marginal benefit and marginal cost of a fiscal program. Spending ministers and individual members of parliament are exposed to political pressures from interest groups and, since taxes fall on the general public while expenditures benefit particular groups, are biased towards large expenditures and large deficits. The prime minister and the finance or treasury minister and broadly-based political parties in parliament, in contrast, do not depend on particular interest groups to the same extent; their decisions are more strongly guided by general economic considerations. The distribution of power between these two groups, therefore, determines the size of the spending bias built into the budgeting procedure.

In view of this, we develop two propositions: (1) Institutions conducive to long-run orientation of fiscal policies enhance fiscal discipline. The basic idea is that long-run orientation mitigates the conflict between shortrun and long-run net benefits. We call this the <u>long-term-constraint</u> <u>hypothesis</u>. (2) Institutions which weaken the role of special interests in the budgeting process are conducive to fiscal discipline. The basic idea is that such institutions mitigate the spending bias arising from the difference between beneficiaries and the general tax payer. We call this the <u>structural</u> <u>hypothesis</u>. We test both hypotheses using fiscal data from the EC member countries of the 1970s and 1980s, and characterizations of the national budgeting procedures obtained from expert assessments.

The main empirical result is a strong support for the structural hypothesis. Specifically, our results suggest that a budgeting process lending the prime or finance (or treasury) minister a position of strategic dominance over the spending ministers, limiting the amendment power of parliament, and limiting changes in the budget during the execution process is strongly conducive to fiscal discipline. In contrast, the role of long-term fiscal constraints in achieving fiscal discipline, while in most cases positive, is not found to be significant. While we do not conclude from this that long-term constraints lack importance, our conclusion is that a long-term constraint alone is insufficient to overcome the problems of fiscal discipline for a country whose budgeting procedure has structural weaknesses.

Our results suggest that institutional reform of the budgeting process is a promising avenue to achieve a larger degree of fiscal discipline. Such reform may be required in some countries to achieve the fiscal targets spelled out recently in the Maastricht Accord, which can be regarded as a special form of long-term constraints on fiscal policies. What is more, institutional reform meeting the requirements and particularities of individual member countries may be a promising route to maintain fiscal stability in the third stage of European Monetary Union as a complement to the imposition of fiscal criteria under the current institutional arrangements.

1. Introduction

The approach of a monetary union in Europe has raised concerns about the appropriate fiscal policy regime in the European Community (EC). One important aspect of this is the link between the degree of fiscal discipline the member governments of a monetary union adopt and the union's long-run inflation rate.¹ Many participants in this discussion fear that a systematic lack of fiscal stability of some members of the European monetary union might create pressures on the European central bank to conduct a too expansionary monetary policy for the Community, with the result of lasting, excessive inflation. Some have even argued that the implicit possibility of taxing citizens of other countries which exists, if the union's central bank can be induced to bail out governments in financial crises, would lead to a deterioration of fiscal discipline in the Community, once the European monetary union is in place. As a result, the revisions of the Treaties of Rome adopted in Maastricht in December 1991 incorporate a procedure to supervise the fiscal performance of the members by the Community and to increase fiscal discipline.

Although the Maastricht Accord calls for the adoption of appropriate budgetary procedures by the member states², the basic approach it takes is one of cotrolling, ex-ante and ex-post, their fiscal performance by the Community. The hope is that, by spelling out budget criteria and procedures and penalties for dealing with violations of these criteria, member governments can be induced to fiscal stability, i.e., long-run budget balance or debt growth not exceeding nominal GDP growth. To be effective, such a strategy must rely on the credibility of the threat it implies for members with deviant fiscal policies. Experience with budget norms in the U.S. suggests that governments find ways to circumvent fiscal restraints in practice, with the result that they are largely ineffective (von Hagen, 1991, 1992). In the European context, the absence of a strict enforcement mechanism of fiscal constraints among sovereign nations other than the threat or

_ 1 _

possible application of peer pressure and financial sanctions raises the problem of how to ensure prospects of budget criteria to be successful.

This study looks at the issue from another perspective. We start from the presumption that budgeting procedures, i.e., the rules according to which budgets are drafted by a government, amended and passed by parliament, and implemented by the government, have important consequences for the degree of fiscal stability attained. In other words, we claim that institutions shape the outcome of the political processes evolving within them. This requires that institutions are fixed relative to the political processes they govern, i.e., the formation of rules and procedures for decision making is not part of the same political process. Instead, the actors are legally bound or have a common understanding that the institutions should be regarded as given. Of course, this does not exclude that the institutions themselves can be changed over time, however, doing so would require a different political process. One variant of this basic claim - that the greater credibility of an independent central bank's commitment to price stability leads to lower inflation rates is, of course, one of the important justifications for the European monetary union itself. Following the same logic, our approach leads to the conclusion that institutional reform of the budgeting process may be a promising alternative for the EC to foster fiscal stability.

The main proposition of this paper is that a budgeting procedure enabling a government to commit itself to fiscal discipline is an essential condition for fiscal stability. Commitment mechanisms are important on all three levels of the budget process, the bargaining within the cabinet of ministers, the passing of the budget law through parliament, and the execution of the budget. Commitment is facilitated by restricting the effect bargaining processes on each level can have on total spending and revenues. It can be provided by formal guidelines determining the outcome of the budgetary process

Fig. 1: General Government Expenditure (perc. of GDP, 1961-90)







or by restricting the scope of changes participants in the bargaining process can make; it can be the result of legal restrictions on the process or of long-standing traditions which are expected, both by the participants in the budgetary process and the public, to be respected in the future (North and Weingast, 1989). We develop and test this proposition in two versions: One focusing on the existence and implementation of long-term fiscal plans and the other focusing on the structural characteristics of the budgeting process. Our empirical results using data from the EC suggest that structural characteristics are important. Specifically, dominance of the prime minister or finance minister over the spending ministers in setting budget parameters, limitations to modifications of the budget proposal by the legislature, and limitations to budget changes during the execution are institutions conducive to fiscal stability.

This study proceeds as follows. Section 2 presents some stylized evidence of fiscal performance in the Community over the past two decades. Section 3 outlines our basic theoretical argument. Section 4 begins with a review of the main characteristics of the budgeting procedures currently used in the 12 countries of the Community based on a questionnaire sent to the member Finance Ministries or Treasuries in 1991. Section 5 presents the empirical tests of our main hypotheses of interest. Section 6 summarizes our main conclusions.

2. Fiscal Performance in the EC, 1971-90: Some Stylized Facts

Figures 1 through 8 give an impression of the fiscal performance of the 12 EC member states over the past decades. Figures 1 and 2 depict the growth of general government expenditures relative to gross domestic product (GDP). Throughout the 1960s, there was remarkable similarity among the six EC members and Ireland, Denmark and the U.K.. Ratios of expenditures to GDP varied

— 5 —

Fig. 3: Net Government Lending (perc. of GDP, 1961-90)



percentage of GDP





between 25 and 35 percent at the beginning of this decade; by its end, they had risen to between 30 and 40 percent. In fact, these ratios generally declined in the late the 1960s. The 1970s brought two important changes: a faster rise of expenditures relative to GDP in most countries - the exceptions being France, and, after initial surges, Britain and Germany - and an increase in the variance of these ratios across the EC members. Expenditure ratios generally peaked in the early 1980s, followed by moderate declines in Germany, France, and the U.K., and more significant reductions in Denmark, Belgium, Luxembourg, Ireland, and the Netherlands. In contrast, Greece, Spain, and Italy maintained positively trending expenditure ratios throughout the decade.

Figures 3 and 4 show the development of net government lending, i.e., total revenues less expenditures, throughout the same period.³ Once again, during the 1960s we find a striking similarity among the European countries. The ratios of net lending to GDP differed by a maximum of about seven percent. Once again, the 1970s brought much larger variation among these countries. After 1975, Italy, Belgium and Ireland had the most rapidly deteriorating budgets. Belgium and Ireland showed some improvement in the mid- and late 1980s, but still retained some of the largest relative deficits in the group. Denmark and Luxembourg are most noteworthy for the wide swings in their net lending ratios during this period. In contrast, France, Germany and the U.K. enjoied steady developments and moderate deficits. With the exception of Italy, Greece and Belgium, the EC countries achieved a greater degree of convergence of net lending ratios again towards the end of the 1980s.

Figures 5 and 6 look at the relative budget balances in terms of net government lending excluding interest payments, called primary net lending for short.⁴ Primary net lending shows to what extent governments accumulate new debt during a period over and above what is required to service existing interest obligations and thus gives a better indication than total net lending









percentage of GDP

of a government's need and willingness to change budget developments. Here, we observe a greater degree of conformity among the 12 countries throughout the whole period. Only Italy and Ireland started the 1970s with primary deficits (net borrowing), only Italy and Greece ended the 1980s in this way. Denmark and Luxembourg are most noteworthy for the large swings in their primary net lending. Ireland's primary deficit deteriorated very rapidly in the mid-1970s, it improved steadily from six percent to a surplus of six percent of GDP between 1981 and 1989. While France and the U.K. had primary surpluses during most of this period, Germany experienced deficits from 1974 through 1982.

Figures 7 and 8 demonstrate the ratios of gross public debt to GDP from 1971 to 1990. At the beginning of the 1970s, the EC was divided in two groups, a relatively high-debt group including Italy, Belgium, the Netherlands, Ireland, and the U.K., with ratios of debt to GDP between 50 and 85 percent, and a relatively low-debt group comprising Denmark, West Germany, France, Luxembourg, Spain, Greece, and Portugal, with ratios between 10 and 30 percent. By the end of the 1980s, three groups are discernable: Luxembourg, France, Spain, the U.K. and Germany, all with debt to GDP ratios of no more than 40 percent; Denmark, the Netherlands, Portugal and Greece, whose ratios are between 60 and 80 percent; and Belgium, Ireland, and Italy, whose ratios are above 80 percent.

Apart from these differences in the debt ratios themselves, the dynamics vary significantly among the 12 countries. The U.K. and Luxembourg consistently experienced falling debt ratios throughout the period, and the French ratio was virtually flat. All other countries experienced significant growth in their debt ratios following the second oil price shock in 1979; onlyIreland and Denmark had significantly rising debt ratios already earlier in the 1970s. Only these two countries managed to reduce their debt ratios significantly during the 1980s, while other countries merely succeeded in

- 11 -





percentage of GDP



Country	1971-75	1976-80	1981-85	1986-90
	Net Lending	; of Government (% of GDP)	
Belgium	-3.54	-6.76	-10.52	-7.10
Denmark	2.94	-1.20	-5.86	0.86
Germany	-1.16	-2.66	-2.42	-1.44
Greece	-	-	-10.16	-15.70
Spain	0.22	-0.32	-5.30	-3.78
France	0.16	-0.64	-2.70	-1.86
Ireland	-6.72	-10.00	-12.00	-6.48
Italy	-7.06	-8.10	-11.48	-10.88
Luxembourg	2.68	2.22	1.24	2.80
Netherlands	-0.74	-2.98	-6.02	-5.70
Portugal	-	-	-10.16	-5.72
υ. κ.	-2.20	-3.80	-3.02	-0.42
EC Average	-1.54	-3.42	-6.53	-4.62
St.D. Median	3.43 -1.16	3.82 -2.91	4.29 -6.02	5.18 -5.70
	Net Governm	ent Lending Excl	uding Interest	
	(% of	GDP)		
Belgium	-0.10	-2.00	-1.02	3.70
Denmark	4.26	1.38	1.90	8.78
Germany	-0.02	-0.94	0.38	1.38
Greece	-	-	-6.32	-7.62
Spain	0.74	-0.76	-3.56	-0.22
France	1.18	0.48	-0.26	1.00
Ireland	-2.28	-4.16	-2.92	2.80
Italy	-4.50	-3.28	-4.16	-2.20
Luxembourg	3.66	3.08	2.28	3.70
Netherlands	2.12	0.22	-0.56	0.36
Portugal	-	-	-3.76	2.32
U.K.	1.58	0.54	1.88	3.50
EC Average St.D. Median	0.66 2.61 0.74	-0.54 2.17 -0.76	-1.51 2.73 -1.02	1.46 3.94 2.32

Table 1: Average Net Government Lending, 1971 - 1990

Country	1971-75	1976-80	1981-85	1986-90
	Gross Public De	ebt (% of GDP)		
Belgium	63.5	68.04	105.08	128.44
Denmark	9.12	23.72	68.68	63.58
Germany	20.10	29.82	40.16	43.82
Greece	22.52	26.34	46.58	77.06
Spain	13.48	15.06	35.58	46.06
France	23.78	23.52	29.26	35.35
Ireland	59.92	71.96	94.68	110.34
Italy	56.88	59.88	72.14	95.30
Luxembourg	21.68	15.28	14.54	10.16
Netherlands	44.54	41.88	60.74	76.26
Portugal	20.13	35.80	56.70	70.50
U.K.	68.56	57.80	58.12	50.40
EC Average St. D. Median	35.35 21.66 22.52	39.09 20.43 29.82	56.86 26.17 56.70	67.27 33.23 63.58

Table 2: Average Gross Public Debt, 1971 - 1990

preventing further increases. The three countries in the high-debt group of 1990 were in the high-debt group of 1971; four of five countries in the lowdebt group of 1990 were in the low-debt group of 1971. Only the U.K. moved from being the high-debt group in 1971 to the low-debt group in 1990. In sum, relative debt performances are very persistent in Europe.⁵

Tables 1, 2, and 3 illustrate the same developments by reporting fiveyear moving averages of net government lending, primary net government lending, gross public debt, and general government expenditure, all expressed as fractions of GDP. The rising standard deviations of the first three variables over this period reflect the increasing disparity in fiscal performance among the 12 European countries. Only the standard deviation of expenditures ratios declines towards the end of the 1980s.

While the preceding graphs and tables demonstrate considerable variation of fiscal outcomes in the EC, they do not tell us anything about the source of these differences. Two extreme scenarios are possible: Fiscal outcomes in individual countries could be completely determined by country-specific, mutually uncorrelated shocks. Alternatively, fiscal outcomes could reflect country-specific responses to the same shock(s). Consider the following twocountry model as an illustration:

$$y_{i,t} = \beta_i x_t + u_{i,t}$$

$$y_{j,t} = \beta_j x_t + u_{j,t}$$

$$cov(u_{i,t}, u_{j,t}) = 0,$$
(1)

where $y_{i,t}$ denotes country i's fiscal outcome variable, x_t is the underlying shock common to both countries, the coefficient β describes a country's reaction to the common shock, and u is its reaction to purely country-specific shocks. The first scenario of purely country-specific shocks implies $x_t = 0$, which in turn implies that $y_{i,t}$ and $y_{j,t}$ are uncorrelated. The second scenario

Country	1971-75	1976-80	1981-85	1986-90	
	Government Expenditure (% of GDP)				
Belgium	42.98	51.32	57.96	53.46	
Denmark	44.36	51.32	60.44	57.66	
Germany	43.16	47.90	48.44	46.32	
Greece	-	-	42.70	49.94	
Spain	23.60	29.18	38.70	41.32	
France	39.12	44.54	50.86	50.48	
Ireland	40.18	45.88	53.30	47.84	
Italy	35.84	39.50	48.38	51.42	
Luxembourg	38.84	51.92	54.58	51.20	
Netherlands	44.96	54.38	60.58	58.88	
Portugal	-	-	44.34	43.50	
U.K.	39.90	42.30	44.66	39.78	
EC Average	39.29	45.80	50.41	49.32	
St. D. Median	6.20 39.12	7.50 45.88	7.14 48.88	5.94 49.94	

Table 3: Average Government Expenditure

Country	Government Expenditure		Net Lending less Interest Paid		Gross Public Debt				
	Variance explained	Fa pa	ctor ttern	Variance explained	Fa pa	ctor ttern	Variance explained	Fa pa	ctor ttern
Belgium	96.5	+	n	99.2	+	+	98.1	+	n
Denmark	94.1	+	-	84.3	+	n	96.4	+	n
Germany	90.0	-	+	92.7	-	n	95.5	-	+
Greece	76.0	n	n	94.5	+	n	83.3	n	-
Spain	96.4	+	-	98.7	+	+	96.3	+	-
France	97.0	-	-	95.6	-	-	96.6	-	-
Ireland	90.0	+	+	95.0	+	+	95.9	+	+
Italy	93.5	+	-	90.9	+	n	97.0	-	-
Luxembourg	85.0	+	n	98.8	-	+	92.9	+	n
Netherlands	93.2	+	-	96.3	+	+	96.7	+	-
Portugal	16.5	n	n	96.4	+	n	20.7	n	n
United Kingdom	91.8	-	+	98.2	-	+	94.9	-	+

Table 4: Factor Analysis of Government Expenditure, Net Lending, and Debt

Note: Variance explained is the percentage of variance explained by two common factors. Factor patterns: '+' or '-' indicates that the estimated coefficient for this factor is above or below the average estimate for all 12 countries. 'n' indicates factor is not significant at the 10 percent level. Period for estimation: 1971 - 1990; 1981 - 90 for Greek and Portuguese expenditure ratios.

holds when $u_{i,t} = u_{j,t} = 0$, and β_i differs from β_j . Note that we can extend equation (1) in a simple way to contain two common shocks:

$$y_{i,t} = \beta_{1,i}x_{1,t} + \beta_{2,i}x_{2,t} + u_{i,t}$$

$$y_{j,t} = \beta_{1,j}x_{1,t} + \beta_{2,j}x_{2,t} + u_{j,t}$$

$$cov(u_{i,t}, u_{j,t}) = 0,$$
(2)

in which case the two outcomes are less than perfectly correlated, even if there are no country-specific shocks, provided that the pairs of reaction coefficients are different between the countries.

To see which of these two paradigms describes the European case best, we apply factor analysis to our ratios of general government spending, net government lending, and gross public debt to GDP. This technique allows to estimate the unobserved common shocks explaining a set of time series, their 'common factors', corresponding to the variables $x_{1,t}$ and $x_{2,t}$ in model (2). By construction, these shocks are uncorrelated with each other and with the country-specific shocks. Having obtained estimates of these factors, we can then use regression analysis to see how much of the variance of the observed fiscal outcomes is explained by reactions to common shocks as opposed to country specific shocks. The results of this procedure are summarized in Table 4. For each variable, two common factors were estimated. The first of each pair of columns indicates the percentage of the total variance of the variable explained by a country's reaction to the two common shocks. For all three variables, we find that almost all the variance is explained by the two common factors. Greece and Portugal are the only two exceptions to this result.⁶ The second column in each pair gives some information about the country's reaction to the common factor as determined by the regression analysis. Here, a '+' means that the regression coefficient on this factor is significant and above the average of the significant estimates for all 12 countries, a '-' indicates a significant coefficient below the average estimate, and a 'n' indicates that the factor was not significant in the regression. The main conclusion from this exercise is that fiscal policies in the EC countries, as described by deficits, spending and debt patterns, can best be characterized by countryspecific reactions to common shocks, rather than responses to country-specific shocks. To the extent that fiscal institutions determine a country's fiscal-

policy response to economic shocks, institutional differences across these countries may be important to explain this variation in fiscal outcomes.

In Table 5, we address the relationship between government expenditures and net lending. According to a popular hypothesis, large and growing deficits - and, therefore, growing debt to GDP ratios - are caused by large and growing levels of government expenditure (see e.g. Roubini and Sachs, 1989a, Larkey et al., 1981). If this was true, greater fiscal discipline could be achieved by reducing government spending. To test this hypothesis, we check for Granger-

Country	R ² of Test Regression	F—Test for Causality	F-Test for Serial Correlation of Residuals
Belgium	0.89	0.16	0.95
Denmark	0.76	0.41	0.56
Germany	0.44	0.19	0.93
Greece	0.83	0.53	0.30
Spain	0.54	0.30	0.41
France	0.53	0.91	0.64
Ireland	0.83	0.08	0.79
Italy	0.87	0.04	0.42
Luxembourg	0.45	0.83	0.80
Netherlands	0.87	0.08	0.93
Portugal	0.95	0.09	n.a.
United Kingdom	0.61	0.33	0.84

Table 5: Causality of Government Expenditure for Net Lending

Note:Tests are for Granger-causality of changes in the expenditure/GDP ratio for net government lending / GDP. Test regression includes two lags of the dependent and two lags of the independent variable. Sample period is 1961 -1990 except for Spain (1971 - 1990), Greece (1980-1990) and Portugal (1980 -1990). Tests for serial correlation of residuals check the significance of four a test regression of the residuals from the causality test regression on four of own lags. All entries are probabilities of F values larger than estimated F's under the Null-hypotheses of non-causality and no serial correlation.

causality of government spending for net lending, both measured relative to

GDP.⁷ Defining d_t as the ratio of net lending to GDP and g_t as the ratio of expenditure to GDP, we estimate the following regression model:

$$d_{t} = \alpha_{0} + \alpha_{1}d_{t-1} + \alpha_{2}d_{t-2} + \gamma_{1}\Delta g_{t-1} + \gamma_{2}\Delta g_{t-2} + e_{t} , \qquad (3)$$

where e_t is a serially uncorrelated regression error. The test for Grangercausality in this context is an F-test on the joint significance of γ_1 and γ_2 .

The first column in Table 5 shows that the test regressions explain most of the variance of the dependent variable. The second one indicates that expenditure ratios do not generally Granger-cause net lending ratios, if standard significance levels are applied. Italy is the only exception.⁸ The last column shows that the regression errors are indeed serially uncorrelated, an important condition for the reliability of the F-test. Overall, Table 5 produces two interesting findings: First, the regression R²'s show that the low-debt countries of 1990, Luxembourg, Spain, France, Germany, and the U.K., have the smallest degree of persistence in net lending ratios in the sense that their past net lending ratios are the least helpful to predict current ratios. This suggests that countries which manage to change their budgets less easily over time are more prone to large deficits and debt. Second, with the exception of Italy, the simple hypothesis that rising deficits are the result of growing expenditures is not warranted by the data.

While tables 4 and 5 compare time series properties of the data across countries, table 6 takes a different perspective and compares average relative performances among the 12 European countries during the 1980s. The table is based on the moving averages calculated in table 1. After ordering the 12 countries according to the size of each variable, we compute the Spearman rank correlation coefficients shown in table 6. The first four coefficients - all significantly positive - confirm and extend the earlier finding that <u>relative</u>

- 21 -

Items	1981-85	1986–90
Net Lending 1981-85 vs. Net Lending 1986-90	0.85 (.0005)	
Net Lending Excl. Interest, 1981-85 vs. 1986-90	0.75 (0.005)	
Gross Public Debt, 1981-85 vs. 1986-90	0.89 (0.00)	_
Government Spending, 1981-85 vs. 1986-90	0.80 (0.001)	—
Net Lending vs. Net Lending Excl. Interest	0.72 (0.002)	0.62 (0.03)
Net Lending vs. Gross Public Debt [.]	-0.82 (0.001)	-0.81 (0.001)
Net Lending Excl. Interest vs. Gross Public Debt	-0.25 (0.43)	-0.14 (0.649)
Net Lending vs. Government Spending	0.01 (0.983)	-0.09 (0.779)
Net Lending Excl. Interest vs. Government Spending	0.57 (0.055)	0.14 (0.665)
Gross Public Debt vs. Government Spending	0.36 (0.245)	0.30 (0.354)

Table 6: Rank Correlation Coefficients

Note: All variables measured as percentages of GDP. Upper entries are Spearman rank correlation coefficients. Entries in parantheses are the probability estimates for t-tests that the correlation coefficient is zero.

rankings in fiscal performances are very persistent over time. That is, a country that was ranked highly in terms of, say, its net lending ratic in the early 1980s was very likely to be ranked highly in the second half of this decade. The fifth row of table 6 indicates that countries with relatively high (low) net lending ratios tend to be countries with relatively high (low) primary net lending ratios. The fifth row indicates that the same is true with regard to net lending ratios and gross public debt ratios.

The following rows of table 6 show that there is no significant

correlation between a country's rank in primary net lending and its rank in gross public debt. That is, there is no systematic link between relatively high (low) primary deficits and relatively high (low) debt levels. Furthermore there are no systematic relationships between relatively large deficits, large primary deficits, or large debt ratios and relatively large expenditure ratios. Once again, this refutes the simple notion that large deficits or large debts are due to excessive spending.

3. Budgeting Procedures and Fiscal Outcomes: Theory

The previous section has demonstrated the significant variation in the fiscal performance of the EC member countries over the past two or three decades. Despite the differences in outcomes it suggests that fiscal policies in the EC were mainly driven by shocks common to all countries. The interesting question then is, what explains the large differences in their reactions?

Economic Models of Fiscal Performance

Conventional economic analysis does not say much about the determinants of fiscal performance. It generally takes fiscal policy as exogenously determined by political processes. One strand of literature tries to explain the secular growth of government expenditures relative to the economy, a tendency first formulated as 'Wagner's Law' (Wagner, 1890). Although the link between economic growth and the relative size of government expenditures, which Wagner attributed to the growing responsibilities of government in the process of industrialization, seems apparent for many countries at first glance, empirical studies generally found no or little support for the 'Law' (e.g. Larkey et al., 1981; Cameron, 1978). Other attempts at generalizing empirical observations have been equally unsuccessful to withstand closer statistical scrutiny.⁹

Another line of research attempts to identify fiscal policy reaction functions linking fiscal variables such as expenditures or deficits to macro economic variables, such as growth or unemployment. Roubini and Sachs (1989a) estimate expenditures and revenues reaction functions for 15 OECD governments. They find that expenditure ratios, on average across the OECD, respond negatively to output growth and positively to rising unemployment rates; revenue-to-GDP ratios respond negatively to output growth and negatively to unemployment rates. They conclude that the slowdown in economic growth and the rise in unemployment after the first oil price shock in 1973 were responsible for the subsequent rise in government expenditures relative to GDP.

Roubini and Sachs's argument would suggest, however, that the rise in expenditures explains much of the deterioration of budget deficits in the 1970s. Earlier, we saw that this simple relationship is not confirmed by our data. Since their estimated reaction functions are the same across countries, differences in the observed outcomes would have to be due to differences in the economic shocks individual governments were reacting to, which is not consistent with our earlier observations, either. Even if different reaction functions were estimated, their approach leaves the question of why such different reactions occurred unanswered.

The same authors (1989b) take a step in a different direction, linking fiscal performance to political characteristics. They argue that government deficits relative to GDP have been largest in OECD countries with relatively unstable governments. More specifically, while the post-1973 slowdown in economic growth and rise in unemployment explain the rise in expenditure and deficits in the OECD, they find that "multi-party coalition governments, especially those with a short expected tenure, are poor at reducing budget deficits" (ibid., p. 922). Roubini and Sachs propose three intuitive

explanations for this finding: The diversity of interests and constituencies of coalition partners, a tendency of coalitions to be status-quo biased, as individual coalition members can block changes from the status quo but cannot organize enough support to push through a change from it, and a lack of enforcement mechanisms for cooperative behavior - which, by assumption, would foster fiscal discipline - in short-lived coalitions.

Roubini and Sachs's argument is, however, not entirely convincing. Their view of unstable governments presupposes that individuals or parties take office for a short time and then disappear from the government sphere. Commitment to longer-run oriented policies does not pay off for such politicians, hence their unwillingness to combat deficits and avoid the accumulation of future tax liabilities through the creation of public debt. However, even if governments change relatively frequently, it may still be true that the members of government are drawn from a pool of candidates or parties which does not change much over time. In such an environment, commitment to longer-run goals does pay off, because government politicians can expect that they or their party will have another turn in the future. While Roubini and Sachs's data suggests that countries with large deficits tend to have unstable government stability than others. This suggests the importance of other, omitted political characteristics.

Political Economy of Budgeting Procedures

Recent politico-economic literature has explored the role of institutional arrangements governing the budgeting process, voting arrangements and commitment mechanisms for fiscal performance. Although this literature is heavily influenced by the peculiarities of the U.S., there are general lessons to infer from this research. The most important one is that institutional structures, i.e., the arrangements assigning the roles individual participants play and the scope and sequence of decisions, have important effects on the final outcomes of the budgeting process. Here, we review some of the main arguments.

In the most primitive form of a budgeting process, budgets would be voted by parliament after a general debate in which each member of parliament submit can proposal. Arrow's (1963) well-known Impossibility Theorem implies that such a procedure does not generally lead to an equilibrium outcome. Only under restrictive conditions on the preferences of the members of parliament does a majority rule induce an equilibrium. Of course, such is not the practice of actual budgeting procedures. In practice, budgeting procedures are divided between government or parliamentary committees drafting a proposal, parliament which may amend the proposal subject to certain restrictions and pass it, and again the executive carrying out the budget law. The politicoeconomic literature focuses on how the specific institutional arrangements affect the existence and the properties of the equilibrium outcome.

Shepsle (1979a, b) distinguishes three characteristics of budgeting procedures. The <u>division of labor</u> arrangement assigns individual actors in the process to specific roles. For example, a typical European arrangement is that government drafts a budget proposal to be presented to the legislature.¹⁰ The arrangement may be that the proposal is drafted by all cabinet members together or in bilateral talks between the finance (or treasury) minister and the various spending ministers. In the U.S., in contrast, the division of labor arrangement is a system of parliamentary appropriations committees. A <u>specialization of labor</u> arrangement is an assignment of jurisdictions to individual groups of actors. A committee may have jurisdiction over only one dimension or several dimensions of government services. Spending ministers usually only have jurisdiction over their own field. Finally, an <u>amendment</u>

- 26 -

<u>control rule</u> specifies what type of amendments the legislature may bring up against the budget proposal. Shepsle considers four types of amendment rules. Under an open rule, all amendments are permissible. Under a closed rule, parliament can only agree or disagree, in which case a fall-back budget is adopted, e.g., the previous one. Under a boundary rule, amendments cannot make budget parameters exceed or fall short of certain limits. Finally, under a germaneness rule, only such amendments can be considered which pertain to the matter currently under discussion.

The importance of the division of labor arrangement is that it determines the agenda setter for the subsequent steps of the budget procedure, i.e., the agent presenting proposals for budget changes over the status quo. This is important, because the proposed changes will depend on the preferences of the agenda setter. If amendments are restricted, the agenda setter can condition the outcome of the process by the choice of a proposal. The specialization of labor arrangement is crucial because it determines what kind of choices the agenda setter can make. In the simplest case, where each committee has jurisdiction over only one budget dimension, its choices are limited to a mere 'more or less' of the relevant variable. In contrast, if the committee has jurisdiction over more than one dimension, its choices can involve trade-offs among the relevant activities. Finally, the amendment control rule determines the power the agenda setter has relative to the legislature. The more restrictive the rule, the more the legislature is bound by the proposal; under an open amendment rule, the situation is equivalent to the primitive budgeting procedure described above.

Shepsle's work generates a number of important insights. First, appropriate institutional structures generate equilibrium outcomes under fairly general assumptions about the agents' preferences, and, in particular, in situations in which no equilibrium exists in the primitive set-up. This

underlines the importance of institutional arrangements for fiscal outcomes. Second, the characteristics of the particular equilibrium outcome depends on the combination of all three institutional characteristics. This implies that different institutional set-ups among the EC countries may explain different reactions to the same underlying shocks. Third, the status quo will be most powerful in determining the outcome, if there is a one-to-one mapping between parliamentary committees and jurisdictions, and if the legislature is bound by a rule that prohibits amendments if the committee proposes the status quo. Thus, regardless of whether or not the government is formed by a stable coalition - which would determine the assignment of individuals to committees or ministries but would not the affect budgeting procedure - institutional structures are important determinants of how likely a deviation from status quo will be.

Division and specialization of labor arrangements together determine the degree of centralization of the budgeting process. In the European context, we distinguish, within government, between a decentralized approach, in which each spending minister with authority over one budget dimension is engaged in bilateral talks with the finance or treasury minister, and a centralized approach, in which the cabinet as a whole discusses the budget proposals. In the parliamentary procedure, the relevant distinction is between a sequence of votes proceeding on an item-by-item basis, or a general vote on the entire budget following a general debate. The importance of this aspect comes from the limits it puts on <u>universalism</u> and <u>reciprocity</u> (Alt and Chrystal, 1981). Universalism refers to the property of budget proposals to 'contain something for everyone', i.e., to distribute favors more generously than an individual decision maker would want. Reciprocity refers to the principle of not attacking another person's appropriation proposal in return for her not attacking one's own. Both tend to increase expenditures.

Chrystal and Alt summarize the basic argument asfollows: Consider a government where each spending minister pursues only his individual interest and the budget law requires a minimal winning coalition within the cabinet. Under such circumstances, it is always rational for a minister to vote against other ministers' proposals for increasing their budgets on cost-benefit grounds. By implication, no minister is able to push an expansion of his budget through and government settles on the cheapest possible budget, although each minister would like a larger one. One way to get around this is to engage in mutual agreements, i.e., proposals which benefit more than one minister, or to agree tacitly not to vote down each others' proposals. The result is that each spending minister obtains a larger budget. Chrystal and Alt argue that pairwise bargaining between the finance minister and the spending minister favors universalism and reciprocity. They base their view on reports that British Cabinet ministers refrain from attacking spending requests from other departments and the observation that the British system does embed pairwise bargaining. Beyond that empirical observation, however, their argument seems implausible. Tacit agreements require monitoring to be effective, which is easier in multilateral bargaining situations than in decentralized ones, because the former give all participants the opportunity to observe each others' behavior. Furthermore, universalism requires the possibility to decide over multiple budget dimensions simultaneously, which would typically not be possible in a decentralized setting. We will hypothesize, therefore, that both practices are more limited in decentralized than in centralized procedures.

Ferejohn and Krehbiel (1987) analyze the importance of the sequence of decisions in the budgeting process for the final outcome. Specifically, they compare a process in which appropriations are voted individually and the overall budget size is a residual with one in which the budget size is voted

- 29 -

first and the structure of the budget is determined afterwards, given the total size. Contrary to popular belief, the latter procedure does not always lead to a smaller budget than the first. When voting on the total size, decision makers anticipate the limits they create for subsequent allocations. Decision makers with strongly skewed preferences in favor of particular budget items are likely to produce larger budgets in the two-step procedure, because they do not accept the implicit need to trade off individual expenditures in the second step. In contrast, decision makers with more balanced preferences are likely to find smaller budget agreements in this way.

Mackay and Weaver (1979) show that, again contrary to popular belief, committees with the power to propose budgets for particular government services do not always propose a larger budget than the median voter would propose, even if the committee members' preference for the particular service are much stronger than the median voter's preference. Applying their argument to Europe, this means that proposals put forth by spending ministers do not necessarily aim at larger spending for their jurisdiction than proposals resulting from a general debate.

North and Weingast (1989) analyze the role of British fiscal institutions introduced after the Glorious Revolution of 1688. The new fiscal constitution resulted in a specific division of labor between the Crown, Parliament, and the Bank of England and created an effective commitment mechanism for the Crown to serve its debt obligations. Comparing public finances in Britain before and after the Revolution suggests that these changes facilitated a more stable and reliable fiscal policy. Bordo and White (1992) compare British and French public finances between 1790 and 1814 and conclude that institutional deficiencies in the French system undermined the credibility of the French government and forced France to conduct significantly less efficient and stable fiscal policies than Britain during
that period.

A Simple Model of the Budgeting Process

According to Wildavsky (1975, p. 4) the budget process is a mechanism through which political interest groups "bargain over conflicting goals, make side-payments, and try to motivate one another to accomplish their objectives". In essence, it is a device for political conflict resolution. In this section, we propose a simple model characterizing the budgeting process as it is found in the EC countries, and derive some hypotheses concerning fiscal discipline.

Our model describes the budgeting process in three stages. On the first stage, government prepares a budget draft to be presented before parliament. The government comprises spending ministers, a finance or treasury minister presiding over financial resources, and a prime minister acting as the chairman.¹¹ Conflicts of interest between the ministers must be resolved in the drafting process. On the second stage, the budget is submitted to parliament, which can amend the proposal and either pass or reject it. We think of this primarily as a bargaining process between government, which now represents a unified position expressed in its proposal, and the parties represented in parliament, which either support or oppose the government. On the third stage, the budget law is executed and further modifications of the law may be possible.

To characterize the process, we assume that taxes are not earmarked for special purposes. Spending ministers are interested in expanding the resources of their own ministries, but indifferent about the resources of other ministries. Their political success is measured in terms of the size of their budgets. In contrast, the prime minister and the finance minister are not bound by particular interests - or not to the same extent - and, therefore, are more constraint by considerations of general public welfare than the spending ministers. For a given amount of total spending, increasing the general tax burden of the economy reduces public welfare.

Spending and financing government programs involve different time horizons. We assume that the benefits from spending money for the majority of government programs are obtained immediately or over a relatively short time period. In contrast, the welfare effects of higher taxes imposed on current tax payers are felt to a large extent only in the medium and long run, since they involve private sector adjustment to changes in tax incentives or net asset returns. In addition, we assume that tax payers are at least partly non-Ricardian, so that deficit financing allows to shift part of the tax burden to finance current expenditures on future tax payers.¹² Finally, we assume that politicians discount the future, so that present or near pay-offs of their actions are weighed more heavily than those in the more distant future.

Spending ministers are inherently biased to push for increased spending of their own ministries, since the resulting taxes or deficits to finance the extra expenditure fall on the general public (or, in the case of deficits, on the general public in the future), while the spending benefits their own constituency and raises their political support. A budget conflict between two spending ministers, A and B, therefore, has the structure described in fig. 9. Here, we compare two basic strategies: small and large expenditures. If both choose small expenditures, the resulting level of taxes and the deficit remain small, if both choose large expenditures, taxes and the deficit are large. Both spending ministers receive the same pay-offs - denoted by the numbers in the upper left corner for A and the lower right corner for B, when they adopt the same strategy. Each would prefer an outcome in which the other chooses the small expenditure size. However, because a minister faces a loss of political support if his colleague reaps a larger allocation within a given budget than

- 32 -

Figure 9:

	Ministry B strategies					
Ministry A strategies	small expenditures	large expenditures				
small expenditures	40 low taxes, small deficit	20 medium taxes, medium deficit				
	40	60				
large expenditures	60 medium taxes, medium deficit	50 high taxes, large deficit				
	20	50				

he does, the equilibrium is the large budget with high taxes and a large deficit. That is, the spending bias works against fiscal discipline.

The prime minister and the finance minister have a larger tendency to limit spending in order to restrain the level of present and future taxation.¹³ From their perspective, the preferred outcome would be that both spending ministers adopt the small-expenditures strategy, especially so in times where the 'status-quo' budget has high levels of spending and taxation and a high deficit to begin with.

An important aspect of the process within government concerns the sequence of decisions determining the size of the budget. In a 'bottom-up' approach, the total size is determined residually after collecting spending requests from all ministries. Alternatively, the cabinet may agree on a general constraint, first, and decide on individual allocations or the structure of the budget given this constraint afterwards. The general constraint may fix the overall size of the budget, total expenditures or the deficit, or consist of a 'golden rule' clause, i.e., the provision that deficits cannot exceed investment or capital expenditures. Constraints fixing only the deficit or a golden rule clause would, however, be less binding for future decisions than total expenditures or the overall size. A general constraint can be strengthened against universalism and reciprocity by giving the prime minister (or the finance minister) the authority to fix the overall size before individual budgets are determined.

Another characteristic of the process within government concerns the participation in decisions. Budget decisions may be reached by the entire cabinet collectively, or, alternatively, through bilateral discussions between the finance minister and each of the spending ministries. Finally, we can distinguish budget processes by the extent to which they connect the current budget to past and future budgets through multi-period budget plans. If multiperiod budget plans exist at all, they may be regarded primarily as a general orientation or as a binding constraint.

The government's budget proposal is submitted to parliament where it becomes subject to another bargaining process..Members of parliament represent local or other constituencies and are, therefore, characterized by a similar if not stronger - spending bias as spending ministers. On the other hand, members of parliament are bound by party discipline. European parties, which are collections of groups of constituencies, are likely to give larger weight to the general interest - as opposed to particular constituencies - in party decisions than individual members of parliament would do in the absence of party discipline. Furthermore, for the members of the party or the parties backing the government in office, party discipline entails voting to support the government, even if the outcome does not fully match the preferences of the individual member of parliament.

Parliament's role is to amend the budget proposal, and to pass or to reject it. While government sets the agenda for the parliamentary debate, its proposal will anticipate parliament's reaction to it. The relationship between government and parliament is characterized, first, by the scope of amendments parliament can consider. In the simplest case, there may be no restrictions on amendments at all. Otherwise, amendments may only be permitted for certain parts of the budget, or parliament may be restricted to amendments proposing increases in expenditures only if they identify the necessary sources of additional finance, or only such amendments that do not (or only negatively) affect the overall size of the budget.

The second dimension of the relationship between parliament and government concerns the political implications of rejecting the budget favored by the government. The strategic effect of the possibility to reject the budget proposal is two-fold. On the one hand, the more likely a rejection leads to the demise of the government, the more it is in government's interest to propose a budget that can be expected to find a solid majority in parliament. This tends to weaken the position of government in the process. On the other hand, members of the parties supporting government in parliament will refrain from proposing changes to the budget proposal if doing so may entail the fall of the government, unless the changes are regarded of outmost importance. This second effect tends to strengthen government's position in the process. While the combined effect is ambiguous, we assume that the latter effect prevails.

Another dimension of the relationship between government and parliament is the quality of information the budget proposal conveys about public finances and government's intentions. A low degree of informativeness allows government to send mixed signals about its fiscal intentions, making parliament's task of monitoring the budget more difficult. At the same time, the informativeness of the budget also determines the degree to which the budget can be monitored effectively by political forces outside parliament, e.g. the media.

Voting procedures within parliament can be characterized by the order and scope of the votes. As within government, they determine the extent to

- 35 -

which reciprocity and universalism can prevail. Parliament may debate and then vote on the entire budget in one step. Alternatively, it may discuss and vote on the budget item by item, possibly followed by a general vote on the budget as a whole. Finally, parliament may first vote on the overall budget size and then debate and vote over the individual items. We conjecture that the latter approach is most conducive to fiscal discipline, while the first approach is most likely to result in large budgets and large deficits.

The third stage of the budget procedure contains the execution of the budget law under the control of government. During the execution new demands for spending or reduced taxation occur in response to unforeseen economic events, as well as discrepancies between planned and actual revenues and expenditures. As in the drafting process, we assume that, for their political interests, spending ministers are more likely to give in to demands for increased spending and more prone to overrunning the limits set by the budget law than the prime minister or the finance minister. Two conflicting forces become important: the degree to which the budget law binds government's actions during the fiscal year, and the degree of flexibility to respond to unforeseen events. How binding the budget law is for government depends on the possibility to propose supplementary budgets during the fiscal year, on the relative importance of open-ended appropriations in the budget, such as social security or unemployment compensation commitments, and on the power of the finance minister to impose spending limits on ministries which exceed their budget norms. The degree of flexibility in the execution of the budget depends on the possibility to transfer expenditures between budget titles, the existence of a budget reserve, and the possibility to carry unused funds forward.

The Long-term Constraint Hypothesis and the Structural Hypothesis

Our model predicts that the structure of interests among spending ministers and members of parliament, patterns of decision making and voting conducive to universalism and reciprocity, and a large degree of flexibility in the execution of the budget all generate a bias for outcomes with large expenditures, high taxes and large deficits. Conversely, institutional arrangements limiting these forces should be conducive to greater fiscal discipline. We pursue this general proposition in two versions. The first version borrows an insight from the literature on time consistency of policy making.¹⁴ We conjecture that policy makers are more inclined to adopt fiscal discipline when they consider the general, long-run trends and consequences of their policies, than when they are engaged in bargaining over details of the budget or in the budget execution in their daily actions. Policy makers who are able to agree on and announce a long-run program of fiscal stability will be tempted to deviate from that program once its implementation is under way, because such deviations serve short-run political interests. This suggests that strong institutions enforcing long-run orientation of fiscal policies are conducive to fiscal stability. This is our

LONG-TERM CONSTRAINT HYPOTHESIS: The more budgetary decisions are tied to a multi-period fiscal program, the greater the degree of fiscal stability achieved.

The second hypothesis focuses on the structure of negotiations and decision making procedures during the budgeting process and the execution of the budget law. We call it the

STRUCTURAL HYPOTHESIS: Budgeting procedures lead to greater fiscal discipline if they give a strong prerogative to the prime minister or the finance minister, if they limit universalism, reciprocity, and parliamentary amendments, and facilitate strict execution of the budget law. In the next section, we develop empirical tests of these hypotheses using data for the European Community.

4. Budgeting Procedures in the EC: A Characterization

The data for the following characterization of the national budgeting procedures in the EC are based on an assessment of information on national procedures provided by the European Commission.¹⁵ This information was available for all member governments, although in varying degrees of detail. No indication of the ranks of the experts within their ministries was made. The advantage of this data base over the alternative of studying each country's relevant legal code is that the information reflect current practices, not simply legal norms. This is particularly important for a country like the U.K., where many of the procedures depend on a common understanding of the actors rather than codified law (Wildavsky, 1975). Of course, this advantage is achieved at the expense of subjectivity in the assessment. Our data base does not generally indicate changes in the national procedures in the past.

Tables A1 - A8 summarize the assessments.¹⁶ Table Al begins with a general characterization of the levels of government in each country. Levels of government vary between two and four. Social security systems are treated as a separate entity within the general government in most countries. The third column of this table characterizes the budgetary status of lower-level governments. Only Denmark and France have regional governments subject to a binding balanced-budget constraint; in Belgium, Luxembourg, and Greece such constraints exist but are not considered to be binding. Lower-level governments in Germany, the Netherlands and the U.K. are subject to a 'golden rule'. In four countries, Belgium, Denmark, Greece, and Italy, regional governments are required to obtain authorization for borrowing from the general government. In four countries, Ireland, Luxembourg, Portugal, and Spain, lower-level governments are autonomous in their budget plans, in all other countries this autonomy is limited, or lower-level governments can be placed under the surveillance of higher-level governments. The final column shows that the central government's budget is drafted, in most member states, by a single ministry, the exceptions being Denmark, Greece, Italy, and Portugal.

Table A2 reviews the existence and strength of multi-annual budget plans. Belgium, France, Greece, Luxembourg, Spain, and Portugal do not specify any multi-annual budget targets at all. However, with the exception of Belgium, these countries do use some form of multi-annual projections as intertemporal guidelines of their budget plans. Germany, Ireland, Italy, and the Netherlands use the overall size of the budget as a multi-period target. The commitment to these targets is political, i.e., it is not binding, but expresses political preferences and the willingness to make significant efforts to come close to these targets. Only in the Netherlands is the multiannual target part of the coalition agreement of the government. Denmark and the U.K. have more specific multi-annual targets, such as total government revenue or expenditures. In Denmark, these targets are only indicative, i.e., they carry less weight than a political commitment. The same is true for revenue and public sector borrowing targets in the U.K., there, however, the spending target is a political commitment.

Table A3 characterizes the rules for preparing the budget draft. Here, the existence of a 'general constraint' indicates that the draft begins with the statement of overall parameters such as total spending, revenues, deficits, or government debt. Belgium, Greece and Portugal operate under no effective general constraint, the Netherlands and Portugal introduced their constraints only recently. Of the remaining countries, Denmark, France, Germany, Luxembourg, and the U.K. observe spending targets in the draft; the same countries except Luxembourg plus Ireland observe a deficit target; France, Ireland and Italy also have constraints on government debt outstanding. Government guarantees to non-government entities are generally not included in the budget; if they are, they enter as a total (France), an estimated spending amount (Italy), or a maximum (Portugal). Seven of the twelve government budgets do not include reserve funds; the French reserve fund is of very limited size. In contrast, Greece, Italy, Portugal, and the U.K. carry reserve funds on the budget. Special funds exist in all member countries, they are usually included in the budget at least to some extent or annexed to it. Finally, the budget is proposed in one comprehensive document in most countries. Exceptions are the U.K. and Ireland, where revenues and expenditures are presented in different documents, while in Italy and Greece separate documents are drafted for different policy domains.

Table A4 reviews the informativeness of the budget. It begins with an overall judgement, ranging from transparent (France, Germany, Greece and the U.K.) to 'hardly transparent' (Italy). Expenditures are generally broken down by function and administrative responsibility; Ireland, Italy, and the Netherlands being the exceptions. Revenues are presented in a breakdown by source in all countries. Only in Germany, the U.K., the Netherlands, and France (though in a separate statement) can expenditure and revenue categories be linked directly to national accounting statistics to facilitate the assessment of their macro-economic effects. Loans extended by the government are generally reported, although in some cases (Germany, Ireland and Greece) only in separate statements. Most countries use a consistent accounting base in their budget documents (either cash or transactions); only Italy and Belgium use mixed accounting bases. Table A5 considers the importance of offbudget activities by looking at the budgetary treatment of special funds at

— 40 —

the government. Greece, Italy, and, until recently, Portugal and Belgium appear to use special funds and off-budget activities extensively. In all other countries, off-budget operations are very limited and special funds are either included in the budget, or reported as part of it.

Table A6 characterizes the voting procedures within government and the legislature as well as the scope of actions the latter has available. An 'agenda', i.e. initial budget guidelines, is set either by the prime minister or the finance (or treasury) minister (Belgium, France, Germany, Greece, Italy) or by the entire cabinet. The agenda may specify overall limits on the budget size (Belgium, France, Germany, Spain, U.K.), limits on spending or deficits (Denmark, Greece, Ireland), or determine specific budget targets together with one of the former. Subsequent negotiations are bilateral between the spending ministries and the finance ministry in Denmark, France, Germany, the Netherlands, Portugal, and the U.K., multilateral in Italy and involve the entire cabinet in Belgium, Greece, Ireland, and Spain.

Parliaments generally do not have the power to make budget proposals. The only exceptions are Germany and Luxembourg, even there, this possibility is of no practical importance. Amendments are subject to various restrictions. Unlimited amendments are possible in Belgium, Denmark, Germany, Greece, the Netherlands and Portugal. In Denmark and Spain amendments must be offsetting in the sense that proposals to increase expenditures in one budget title must be accompanied by proposals to reduce expenditure in other titles, or that a proposed tax reduction must be matched with an increase in another tax. The French rule is even stricter, prohibiting proposals to raise expenditures in one title and to reduce expenditures under another title. Only in Greece, Italy, and Spain amendments are not perceived to be a potential cause for the government to fall.

Parliaments in Belgium, Luxembourg and Spain vote on expenditures and

- 41 --

taxes simultaneously. In France, existing entitlements are voted upon in a first, general vote, the subsequent discussion on new authorizations proceeds chapter by chapter. In most EC countries, the parliamentary debate ends with a final, global vote on the budget. Only in France and the U.K. such a vote is taken before the parliamentary debate begins. In the Netherlands, a general discussion on the revenue and expenditure sides of the budget opens the parliamentary process, but no vote is taken. Most parliaments are subject to explicit time limits by which the budget must have been passed. Exceptions are only Ireland, the Netherlands, and Spain. Provisional budgets are implemented if the time limit is not met. In most cases, the provisional budget is a prolongation of the previous budget on a monthly or four-months basis.

Table A7 reports the budget monitoring rules during the execution process. Supervision is generally the task of the finance ministry; only in Denmark and Germany are the spending ministers responsible for monitoring their budgets. The French case is an intermediate one, where the finance ministry places supervisors in the spending ministries. Parliaments are generally informed about the execution on a monthly or quarterly basis, but have no further role in the process. In all countries except Belgium, the final closure of the budget occurs during the year following the execution.

Table A8 summarizes the provisions governing the execution of the budget law and, in particular, the degree of flexibility governments have at that stage to react to unforeseen events or deviate from earlier plans. Finance ministers in France, Germany, Greece and Luxembourg can block expenditures if budget overruns are foreseen. In these countries and Denmark, Portugal and the U.K. spending ministries are subject to cash limits during the year, imposing constraints on the timing of expenditures. Only in Belgium, France and Germany, spending ministers generally have to obtain authorization for actual disbursements; in the Netherlands and Portugal, this may be the case

occasionally.

There are various degrees of limitations to transfering expenditures between chapters of the budget during the execution. Transfers often require authorization by the finance minister or the parliament. The German and the Greek finance minister, and the Dutch government with some limitations, may authorize larger budget changes; in all other countries, a new budget law is required and must be passed under the same rules as the original one. In Belgium and Italy, the submission of supplementary budgets during the fiscal year is a standard practice (in March or in June and before October, respectively). Carrying-over unused spending authorizations to the next fiscal year is generally allowed in Italy, impossible in Greece and Ireland, and subject to limitations in the other countries.

Construction of the Long-term Constraint and the Structural Indexes

To test our hypotheses about the effect of institutional arrangements on fiscal outcomes, we need numerical representations of the institutional characteristics of the 12 countries. Such representations can be derived by constructing indexes ranking the budgeting procedures according to the relevant institutional properties. For this purpose, we select those characteristics from tables A1 - A8, which generate the largest differences between institutional arrangements. Considering the remaining ones would not add more discriminating information to our indexes, but simply increase the index value for all countries. We group these characteristics under five larger items: the structure of negotiations within government, the structure of the parliamentary process, the informativeness of the budget draft, the flexibility of the budget execution, and the longterm planning constraint. For each characteristic, we use numbers ranging from zero to four to describe its quality, with a low number indicating a quality conducive to a small degree of fiscal discipline. This assures that each characteristic will contribute equally to the overall indexes.¹⁷ Where the available information was not explicit about the relevant sub-item for a country, we assign a number equal to the average of the available numbers for the other characteristics of the same item. See the appendix for more details.

Next, we construct two sets of indexes. The structural indexes, SI1 -SI3 pertain to the structural hypothesis formulated above. A large structural index SI1 signals the following properties of a country's budgeting procedure: A strong position of the prime minister or finance minister in government and government negotiations evolving under a firm general constraint on the size of the budget; a parliamentary process with strong limits on amendments, votes proceeding item-by-item on expenditures and a global vote on the total size of the budget preceding the parliamentary debate; a large degree of transparency of the budget; and an execution process with limited flexibility and a strong position of the finance minister vis-a-vis the spending ministers. Our structural hypothesis says that countries with a large structural index should be expected to exhibit a relatively large degree of fiscal discipline.

Computing the index SI1 by summing up the sub-indexes for the individual characteristics presumes that these characteristics are substitutes for each other in achieving the same degree of fiscal discipline. To see how important this substitutability assumption is, we compute two more structural indexes by leaving out components from the overall index SI1. Specifically, index SI2 drops the informativeness characteristics. Index SI3 drops these and the flexibility characteristics of the budget execution. Thus, the finding that SI1 is strongly related to fiscal discipline while SI2 is not would indicate that the informativeness aspect is very important relative to the other aspects. If SI1 and SI2 relate equally to fiscal discipline, the informativeness aspect would be less important than the remaining







SI3

SI2

SI1

characteristics.

To test the importance of the long-term constraint, we construct the indexes CON1 - CON3 in a similar matter. We conjecture that, apart from the existence of a multi-annual fiscal program, a high degree of informativeness of the budget facilitating its monitoring, limited amendment power of parliament and limited flexibility in the budget execution are important to make the long-term constraint effective. Thus, a large value of CON1 indicates the existence of a multi-annual fiscal target which is seen as a strong political commitment and is embedded in a consistent economic forecasting framework; a large degree of transparency of the budget; limited amendment power of parliament and limited flexibility of the budget execution. A large value of the index should signal a relatively high degree of fiscal discipline. As before, we construct alternative indexes to assess the implied substitutability assumptions. CON2 leaves out the transparency characteristics, while CON3 leaves out transparency and amendment limits. Graphical representations of all six indexes are shown in figures 10 and 11.

5. Budgeting Procedures and Fiscal Performance: Empirical Tests

We now turn to testing our two main hypotheses using nonparametric tests and regression analysis. While linear regression analysis is more familiar and easier to interpret, nonparametric tests have the advantage of not requiring the explicit specification of a functional form of the relationship between fiscal outcomes and our indexes characterizing budgeting procedures, which is necessarily largely arbitrary. Furthermore, the nonparametric tests have more power in the small sample size we are confronted with.

Table 7 presents the Spearman rank correlation coefficients between our indexes and gross debt ratios, net lending ratios and primary lending ratios for the first and the second half of the 1980s. Rankings for the debt and net

lending ratios assign high ranks to relatively small debt ratios and small deficits; ranks for the indexes are high for relatively large indexes. Table 7 shows that there are strong positive rank correlations between the structural indexes and net lending ratios throughout the 1980s. The rank correlations between the structural indexes and the primary net lending ratio are strongly positive in the first half of the 1980s; they maintain the anticipated negative sign in the later 1980s, but are not large enough to be significant. The rank correlations between the structural indexes and the gross debt ratio have the anticipated positive sign in both subsamples, but they are significant only in the second one. Over all, these results indicate that countries ranking high on the structural index - countries which, under the structural hypothesis should exhibit relatively strong fiscal discipline rank high in debt ratios and net lending ratios. This is a first empirical support for our structural hypothesis.

Table 7 does not, however, support the long-term constraint hypothesis. No rank correlation between the longterm constraint indexes and the debt and net lending ratios are significant; some even have the wrong sign.

Table 8 presents the results of two nonparametric tests for the structural index. To perform these tests, we divided our sample into the group of four countries with the largest values of the structural index, another group of four countries with the four smallest values, and the group of the remaining countries. The Null-hypothesis for these tests is that the distribution of debt and net lending ratios is the same in all three groups, hence the same as in the combined group, indicating that classification according to the indexes does not matter. To alleviate the degrees-of-freedom problem, we perform these tests not only for the two subsamples of the 1980s, but also for a combined sample, assuming that the effect of the budgeting procedures is the same in both subsamples. Thus, the tests under the column

'81-90' use 24 observations of the debt and net lending ratios, two for each country.

	Gross De	ebt/GDP	Net Lend	ling/GDP	Net Lending Excl. Interest/GDP				
	F	period	F	period	pe	eriod			
	81-85	86–90	81-85	86–90	81-85	86–90			
	Structural Hypothesis								
SI1	.42	.60**	.68**	.61**	. 57**	.15			
S12	.26	. 50*	.56*	.58**	.63**	. 32			
SI3	.37	. 60**	.67**	.71***	.71***	.26			
		Long-term	Constrair	nt Hypothesis					
CON1	.15	. 38	. 39	. 37	. 31	19			
CON2	- 11	06	11	24	25	01			
CON3	26	10	03	.12	.12	02			

Table 7: Rank Correlation Coefficients

Note: Table entries are Spearman rank correlation coefficients between the structural indeces and the fiscal performance variables and the longterm constraint index and the fiscal performance variables, respectively. *, **, and *** indicate that the correlation coefficient is significantly different from zero at the ten, five, and one percent significance level.

Table 8 strengthens the evidence shown in table 7. We find that the structural index has a significant impact on the debt ratio, the net lending ratio, and the ratio of net lending excluding interest payments to GDP. As in table 7, the results are less strong when the two subsamples are considered separately for the debt ratio and the ratio of net lending excluding interest payments, however, in the combined sample, the tests are strongly significant for both. The implied relationship - not visible in the table - is as anticipated under the structural hypothesis: Countries with high index values have debt ratios (net lending ratios) significantly more concentrated around

	Gross	Gross Debt/GDP			Net Lending/GDP			Net Lending Excl. Interest/GDP		
	period				period			period		
	81-85	86–90	81-90	81-85	86–90	81–90	81-85	86–90	81-90	
	Kruskal - Wallis Test									
SI1	.12	.02	.00	.03	.02	.00	.12	. 53	.01	
SI2	.74	.17	.16	.15	.05	.01	. 05	.26	.00	
SI3	.06	.02	.00	.03	.02	.00	.09	, 70	.01	
				van de	r Waerde	en Test				
SI1	.11	.03	.00	.03	.03	.00	.12	.45	.01	
SI2	.69	. 20	. 17	, 20	, 06	.01	.06	. 24	.00	
SI3	.06	.02	.00	.03	.03	.00	.10	, 60	.09	

Table 8: Hypothesis Tests for Structural Index

Note: Both tests are Chi-square distributed under the Null-hypothesis of no effect from bargaining and voting structure. Table entries are the marginal probabilities of a larger than estimated test statistic under the Null.

	Gross Debt/GDP			Net Lending/GDP			Net Lending Excl. Interest/GDP			
	period			period				period		
	81-85	86-90	81–90	81-85	86-90	81-90	81-85	86–90	81-90	
	Kruskal — Wallis Test									
conl	.47	. 39	. 20	. 20	.47	.11	.40	. 87	.15	
con2	.79	.78	.72	.94	.69	. 69	.67	.78	.43	
con3	.66	.44	.25	.16	. 35	. 04	, 11	.46	.01	
				van de	r Waerde	en Test				
conl	. 52	.44	.26	.21	. 57	.19	. 50	.94	.26	
con2	.80	.81	.72	.99	.75	.83	.73	.90	. 54	
con3	. 68	.49	.28	.15	.35	.04	.13	.49	.01	

Table 9: Hypothesis Tests for Long-Term Planning Constraint

Note: Both tests are Chi-square distributed under the Null-hypothesis of no effect from longterm planning contraint. Table entries are the marginal probabilities of a larger than estimated test statistic under the Null.

lower values (around zero) than countries with high low index values. The results for the gross debt ratio, where SI1 and SI3 are significant, while SI2 is not, suggest that the characteristics pertaining to the bargaining and voting structures in government and parliament and the flexibility of execution are more important to determine debt ratios than the informativeness of the budget. In contrast, the table indicates that all three indexes are important for the net lending ratios.

Table 9 presents the corresponding test results for the indexes of longterm constraint. Here, we find only weak evidence that the constraint is effective: Only for the net lending ratios and the index CON3, which comprises the characteristics of the longterm planning constraint and the flexibility of execution, and only in the combined sample do the tests indicate a significant role of the longterm constraint in shaping fiscal outcomes.

Table 10 presents the results of estimating the linear regression equation:

$$y_i = a_0 + a_i x_i + u_i,$$
 (4)

where y_i is country i's debt ratio, net lending ratio, or primary net lending ratio, x_i is country i's structural index SII_i or longterm constrain index CONI_i, and u_i is a regression error. Table 10 presents the results of regressions for the combined subsamples. Additional regressions testing for differences in the coefficients between these two subsamples indicated no parameter change and are not reported. The results for all three fiscal variables and the structural index corroborate our earlier findings. The structural index has a significant positive impact on the net lending ratios and a significantly negative impact on the debt ratio. These results can be interpreted as follows: Implementing a structural reform of the budgetary process in a country which increases the structural index by 25 points (approximately the difference between Belgium and the Netherlands, would result, in the long run, in an increase in the net lending ratio by about 4.75

Dependent Variable	Const. (t—ratio)	SI1 (t-ratio)	CON1 (t-ratio)	R ²	RMSE	F				
			Net Lending/GDP							
NLi	-12.72	0.19		0.34	3.94	0.003				
	(-5.64**)	(3.39**)								
NLi	-11.88		0.21	0.19	4.38	0.034				
	(-4.06**)	1	(2.26)							
NLi	-12.26	0.20	-0.05	0.35	2.52	0.04				
	(-4.56**)	(2.25*)	(-0.33)							
	Net Lending Excl. Interest / GDP									
NLX _i	-5.33	0.10		0.33	2.28	0.004				
	(-4.09**)	(3.28**)								
NLX _i	-4.81		0.12	0.18	2.52	0.04				
	(-2.86**)		(2.17)							
NLX _i	-5.04	0.12	-0.03	0.33	2.32	0.01				
	(-3.24**)	(2.22*)	(-0.37)							
			Gross Debt	= / GDP						
B _i	98.97	-0.93		0.24	26.6	0.02				
	(6.50**)	(-2.60**)								
B _i	89.86		-0.94	0.09	28.94	0.14				
	(4.65**)		(1.51)							
B _i	92.27	-1.28	0.69	0.25	26.90	0.04				
	(5.12**)	(2.11*)	(0.72)							

Table 10: Regression Estimates

Note: * and ** indicate significance of the t-ratio at the five and onepercent levels, respectively. F is the marginal probability of a larger estimate of the F-ratio under the Null-hypothesis that the model has no explanatory power. RMSE is the root mean squared error.

percent, and the primary net lending ratio by about 2.5 percent, while reducing the debt ratio by about 23 percent. Additional regressions for the structural indexes SI2 and SI3 yield similar results and are not reported. They did not indicate major differences in the importance of the subindexes.

The regression results for the longterm constraint index are very different. Here, the only significant coefficient is found in the model for the net lending ratio. The index does not, in contrast, explain variation in the gross debt ratio nor the ratio of net lending excluding interest payments. Furthermore, the third regression for the net lending ratio, which combines both the structural index and the longterm constraint index, suggests that the structural index has more explanatory power, since the longterm constraint index is not significant in this regression.

6. Conclusions

This paper investigates the role of budgeting procedures in determining a country's fiscal performance. Recent theoretical literature has developed models showing that the institutional framework in which budgeting takes places can have important consequences for the degree of fiscal discipline achieved by a country. We analyze this basic proposition in two versions, one focusing on structural characteristics of the budgeting procedures and the other focusing on the importance of a longterm fiscal constraint. The empirical analysis uses data for the European Community countries during the 1980s and characterizations of their budgeting procedures focusing on current practices rather than simply on legal norms.

Our main result is that we find strong empirical support for our structural hypothesis. Specifically, our results suggest that a budgeting process that gives the prime or finance (or treasury) minister a position of strategic dominance over the spending ministers, that limits the amendment power of parliament, and that leaves little room for changes in the budget during the execution process is strongly conducive to fiscal discipline, i.e.,

— 53 —

relatively small deficits and public debt. This result can be interpreted as pointing to the importance of a commitment mechanism in the budgeting process. Spending ministers are exposed to political pressures from interest groups and, since taxes fall on the general public while expenditures benefit particular groups, are biased in their decisions towards large expenditures and large deficits. The prime minister and the finance or treasury minister, in contrast, do not depend on particular interest groups to the same extent; their decisions are more strongly guided by general economic considerations. A strong position of the prime minister or the finance (or treasury) minister and limited parliamentary power enable the government to commit fiscal strategies limiting expenditures and deficits and to defend these strategies against the political pressures for more profligate policies which spending ministers and members of parliament are exposed to in the day-to-day political process.

In contrast, the role of long-term fiscal constraints in achieving fiscal discipline, while in most cases positive, is not found to be significant. While we do not conclude from this that long-term constraints lack importance, our conclusion is that a long-term constraint will only be effective if governments have an effective commitment mechanism in the budgetary process. That is, a long-term constraint may improve the fiscal performance of a country with budgeting procedures which rank high on our structural index, but the long-term constraint alone is insufficient to overcome the problems of fiscal discipline for a country that ranks low on the structural index.

Our results suggest that institutional reform of the budgeting process is a promising avenue to achieve a larger degree of fiscal discipline. Of course, this is not to say that institutional reform can discipline political actors who are largely unwilling to accept less spending, higher taxes, or

smaller deficits. Our view is, however, that appropriate institutional reforms can help governments and parliaments to materialize a newly reached consensus for greater fiscal discipline. Such reforms may be required in some countries to achieve the fiscal targets spelled out recently in the Maastricht Accord, which can be regarded as a special form of longterm constraints on fiscal policies. What is more, institutional reforms meeting the requirements and particularities of individual member countries may be a promising route to maintain fiscal stability in the third stage of European Monetary Union as a complement to the imposition of uniform fiscal criteria under the current institutional arrangements.

References

- Alt, James, and K. Alec Chrystal (1981), "Electoral Cycles, Budget Controls, and Public Expenditure". <u>Journal of Public Policy</u> 1, 37-59
- Barro, Robert F. (1974), "Are Government Bonds Net Wealth?" <u>Journal of</u> <u>Political Economy</u> 82, 1095 - 1117
- Bordo, Michael, and Eugene White (1991), "British and French Finance During the Napoleonic Wars". Working paper, Rutgers University
- Cameron, David R. (1978), "The Expansion of the Public Economy: A Comparative Analysis". <u>American Political Science Review</u> 72, 1243 - 1261
- Dearden, James A., and Thomas A. Husted (1990), "Executive Budget Proposal, Executive Veto, Legislative Override, and Uncertainty: A Comparative Analysis of the Budgetary Process". <u>Public Choice</u> 65, 1-19
- European Commission (1983), "Budgetary Systems and Procedures for the Central Government Budget in the Community Member States". <u>European Economy</u> 15, 87-123

(1989) "National Budgetary Systems and Procedures in the Community". Internal Doc. II/266/89-EN-REV

- Ferejohn, John, and Keith Krehbiel (1987), "The Budget Process and the Size of the Budget", <u>American Journal of Political Science</u> 31, 296-320
- Fratianni, Michele, and Jürgen von Hagen (1992), <u>The European Monetary System</u> <u>and European Monetary Union</u>. Boulder: WestView
- Kydland, Finn E. and Edward S. Prescott (1977), "Rules Rather Than Discretion: The Inconsistency of Optimal Plans". Journal of Political Economy 85, 473 - 91
- Larkey, Patrick K., Chandler Stolp, and Mark Winer (1981), "Theorizing about the Growth of Government", <u>Journal of Public Policy</u> 1, 157-220
- Mackay, Robert J., and Carolyn L. Weaver (1979), "On the Mutuality of Interests Between Bureaus and High-Demand Review Committees: A Perverse Result". <u>Public Choice</u> 34, 481-91
- Neumann, Manfred J. M. (1991), "Precommitment by Central Bank Independence" in: <u>Open Economies Review</u> 2, 95 - 112
- North, Douglass C. and Barry R. Weingast (1989), 'Constitutions and Commitment: The Evolution of Institutions Governing Public Choice in Seventeenth-Century England' Journal of Economic History 49, 803-31
- OECD (1987), <u>The Control and Management of Government Expenditure</u>. Paris: Organization for Economic Co-operation and Development
- Roubini, Nouriel, and Jeffrey Sachs (1989a), "Government Spending and Budget Deficits in the Industrial Countries". <u>Economic Policy</u>, 100-127

(1989b), "Political and Economic Determinants of Budget

Deficits in the Industrial Democracies". <u>European Economic Review</u> 33, 903-38

- Shepsle, Kenneth A., (1979a), "Institutional Arrangements and Equilibrium in Multidimensional Voting Models". <u>American Journal of Political Science</u> 23, 26-59
- (1979b), "The Role of Institutional Structure in the Creation of Policy Equilibrium". in: Douglas W. Rae and Theodore J. Eismeier (eds.), <u>Public Policy and Public Choice</u>. Sage Yearbooks in Politics and Public Policy Vol. 6, Beverley Hills: Sage
- von Hagen, Jürgen (1991), "A Note on the Empirical Effectiveness of Formal Fiscal Restraints". Journal of Public Economics 44, 99-110
 - (1992), "Fiscal Arrangements in a Monetary Union Evidence from the U.S.". in: Don Fair and Christian de Boissieu (eds.), <u>Fiscal</u> <u>Policy. Taxes. and the Financial System in an Increasingly Integrated</u> <u>Europe</u>. Deventer: Kluwer
- and Michele Fratianni (1991), "Monetary and Fiscal Policy in a European Monetary Union: Some Public Choice Considerations". in: Paul J. J. Welfens (ed.) <u>European Monetary Integration — From German Dominance</u> <u>to an EC Central Bank</u>. Berlin: Springer

Wagner, Adolph (1890), Finanzwissenschaft Vol. 2, Leipzig: Winter

Wildavsky, Aaron (1975), <u>Budgeting: A Comparative Theory of Budgeting</u> <u>Processes.</u> Boston: Little, Brown & Company

NOTES

1. For review of this discussion see von Hagen and Fratianni (1991), Fratianni and von Hagen (1992).

2. Art. 3 of the Protocol on the Excessive Deficit Procedure requires that 'the Member States shall ensure that national procedures in the budgetary area enable them to meet their obligatins in this area deriving from this Treaty'.

3. Net government lending corresponds to the negative of the general government deficit.

4. Net lending excluding interest corresponds to the negative of the 'primary government deficit'.

5. More specifically, we can calculate the transition probabilities from 1971 to 1990 as follows:

position in 1971	position in 1990						
	high	middle	low	total			
high	3	1	1	5			
low	0	3	4	7			
total	3	4	` .5	12			

Thus, the probability of being in the high-debt group in 1990 for a country being in the high-debt group in 1971 is 60 percent, the probability of being a low-debt country in 1990 for a low-debt country in 1971 is 4/7.

6. Note that the Greek and Portuguese data start only in the 1980s. These countries could therefore not be included in the estimation of the common factors without severe losses of degrees of freedom.

7. A variable x is said to Granger-cause a variable y, if including past values of x together with past values of y in a forecast model for y reduces the variance of the prediction error compared to a forecast model that uses past values of y alone.

8. In addition, there is some weak evidence for Granger-causality for the Netherlands, Ireland, and Portugal.

9. see Larkey at al. (1981). Among these are the 'Displacement Effect Hypothesis' claiming that the relative expansion of government expenditures is driven by the development of government revenues, which increase relative to GNP in times of social turmoil and are rigid downwards, afterwards, and hypotheses about the uncontrollability or stickiness of government expenditures following severe macro economic shocks.

10. Here and subsequently we use the term government in the narrow sense of the cabinet of ministers.

11. Subsequently, we use the term finance minister exclusively to describe this role.

12. Tax-payers are said to be Ricardian if they do not regard public debt as part of their net wealth and, hence, reduce current consumption one-for-one with an increase in government deficits. Ricardian tax-payers would carry the full tax burden of a given amount of government expenditure regardless of whether it is tax-financed currently or deficit-financed and the taxes (plus interest) are collected later. see e.g. Barro (1974).

13. To the extent that a large budget conveys a powerful government, they may have a preference for enlarging the budget size. We assume that this incentive is weaker than their concern for general interests.

14. see Kydland and Prescott (1977) for a classical exposition of the time consistency problem in economic policy. In the monetary policy context, the time consistency problem - generally expressed as the problem of central bank credibility - has been shown to be an important source of persistent inflation and is an important justification for central bank independence. see Neumann (1991) or Fratianni and von Hagen (1992).

15. This data was partly obtained from Member States concerning their situation in 1991.

16. The following tables compress and summarize the assessment of information on national budgeting procedures and are prepared by the present author. They do not reflect the judgement nor the interpretation of the European Commission.

17. Note that this normalization implies that some index values are not integers.

Appendix: Budgeting Rules in the European Community Countries

Characterizations and Index Construction

Abbreviations:

- B: public debt
- D: Deficit
- Depts: Departments
- G: Government spending
- lim: limited
- MF: Ministry of Finance (or Ministry of Treasury where appropriate)
- Parl: Parliament
- T: Tax revenues
- Y: nominal GDP
- ΔB: net borrowing
- "Golden rule" refers to the provision that the budget defict must not exceed investment or capital expenditure.
- <u>Sources</u>: Assessment of information on national procedures obtained from the European Commission in 1991, OECD (1987), and European Commission (1983).

Country	Levels of Government	Budgetary Authoritie	Status of Reg s	Ministries involved in draft of overall balance	
		Balanced Budget Required ¹	Borrowing Author- ized ²	Planning Autonomy ³	
Belgium	3 + Social Security	У	у	S	Finance, Budget
Denmark	3 + Social Security	b	У	S	Finance, Economic Affairs, Revenues
France	4 + Social Security	b		lim	Finance
Germany	3 + Social Security	g⁴	n	S	Finance
Greece	2 + Public Entities	у	У	n	Finance, National Economy for investment
Ireland	2	n	n	У	Finance
Italy	3	n	у	lim	Finance, Treasury, Budget
Luxembg.	2 + Social Security	у	n	У	Budget
Netherlands	3 + Social Security	8	n	S	Finance
Portugal	2 + Social Security	n	n	у	Budget, Fiscal Affairs, Treasury ⁻
Spain	4 + Autonomous Regions	n	n	У	Finance
United Kingdom	2	8	n	lim	Treasury

Table A1: Structure of General Government

¹ n: no requirement; y: requirement exists but is not considered binding; b: requirement is binding; g: 'golden rule' requirement exists, i.e., deficits must not exceed investment expenditures.

² y: Authorization from higher level government is required; n: no requirement.

⁹ y: lower-level governments are autonomous; s: they may be placed under surveillance of higher-level government; lim: they have limited autonomy; n: they have no autonomy.

⁴ 'Golden Rule' applies, i.e., deficits must not exceed investment expenditures.

Country	Target	Nature	Period	Degree of Commitment
Belgium	none			
Denmark	G, I	Estimates in current prices reflect long-term policies as framework for planning. Targets change with political priorities, demographic changes.	t+3	indicative, base for spending limits
France	none	Three-year projections of main items serves to clarify budget decisions. Based on macro- economic scenarios revised several times a year.	t+2	unpublished, for internal use only
Germany	total budget size	Framework for orientation of all interested parties. Annually adapted to current economic situation on the basis of macroeconomic projection.	t+4	political
Greece	none	Starting in 1990, budget is seen as part of 5- year economic plan approved by Parliament. For 91 - 93, budget is part of 3-year stabilization program.		indicative, though in context with EC loan
Ireland	total budget size	Estimates of fiscal developments assuming no changes in policies. Baseline for decisions. Projections are based on macroeconomic forecasts and 'known' factors. Spending departments must provide annual forecast of resource requirements with demands for funds.	t+4	unpublished, but political
Italy	total budget size	2 versions: one on a no-change-in-policies basis and one based on revisions of medium-term fiscal program. The latter contains macroeconomic projections and deficit targets.	t+3	political
Luxembg.	none	multi-annual plan in preparation		
Netherlands	total budget size	Based on macroeconomic forecast updated bi- annually. Interpolations are made in between updates in light of new commitments and policies.	t-1 to t+4	strong political, part of coalition agreement
Spain	none	ad hoc macroeconomic and budgetary scenarios.	5 years	unpublished internal orientation
Fortugal	none	Budget includes indicative projections for programs and projects. For 1991–95, there is a reference framework for medium-term fiscal stabilization and adjustment to the EC average.	t+4	indicative, currently political for D/Y
U.K.	G, Τ, ΔΒ	Medium term fiscal policy objectives for the year ahead are based on macroeconomic forecast. Following years are based on consistent assumptions and Government's inflation objective.	t+4	indicative for budget balance, political for G

Table A2: Multi-Annual Budget Plans or Targets

.

Country	general constraint	government guarantees included	reserve funds included	special funds included	budget in one document
Belgium	double norm on expenditure and deficit (avoidable)	only borrowing of parastatal and government funds	none	since recently most included	recently yes
Denmark	G, D	no, only survey of existing	none	most included	yes
France	D/Y, G, B/Y	as total without specified limits	yes, but very limited	included	yes
Germany	Golden rule,targets from multi- annual plan, sometimes more specific	guarantees to sectors outside general government not included	none	no, but some Sondervermögen annexed to budget	yes
Greece	D/Y, but not observed	not all	yes, at Finance Ministry	annexed	no ¹
Ireland	B/Y, D/Y	no	no	several	no
Italy	B/Y, D/Y	yes, as estimated spending	general and specific	at Interior and Defense Min.	no²
Luxembg.	G/Y		no'		yes
Netherlands	recently D/Y, B/Y	guarantees to sectors outside general government not included	none	included	yes
Portugal	recently	only annual maximum	several	no ⁴	yes
Spain	none		none	most included	yes
U.K.	G/Y, D from multi-annual plan	no	yes	yes	no

Table A3: Rules for Preparation of Budget Draft

¹ Public investment budget is prepared by the Minister of the Economy alone, Finance Minister has no say in it.

² Budget law reflects current legislature and cannot introduce new taxes nor expenditure. New legislation affecting the budget is introduced in the Finance Bill.

³ There exists a rederve fund, which, however, is not available for contingences. The reserve fund receives annual budget surpluses and is used to finance investment in the following year.

⁴ Special and autonomous funds are reported in the budget, but not voted on by Parliament.

the second s							Contraction of the local division of the loc	
Country	Overall assessment	E x F u c t	E x P A d m i n	R e V S o u r c e	N a t A c c	C a p i t a 1	L o a n s	A c c t B a s e
Belgium	not always transparent	У	у	У	n¹	У	у	m
Denmark	incomplete, but transparent	у	У	у	Р	у	у	t
France	transparent	У	у	у	ο	у		t
Germany	fully transparent	у	у	у	у	у	ο	
Greece	transparent	у	у	у	g	у	ο	
Ireland	incomplete, not fully transparent	n	У	у	n	у	ο	с
Italy	hardl y transparent	n²	n ³	У	n	у	У	m
Luxembg.	not fully transparent	У	у	у	р			
Nether- lands	incomplete, not fully transparent	У	n	у	у	у	у	с
Portugal	not fully transparent	У	у	у	p	у	n ⁴	с
Spain	partially transparent	У	у					с
U.K.	transparent	y	у	у	у	у	у	с

Table A4: Transparency of the Budget

Notes: Exp. Funct.: breakdown of expenditures by functions; Exp. Admin. : breakdown of expenditures by administrative responsibility; Rev. Source: Breakdown of revenues by source; National Acc.: link to national accounts established; Capital: capital expenditures identified; Loans: Loans of government reported; Acct. Base: Accounting base; c: cash basis; g: National account codes provided; m: mixed accounting base; o: provided separately p: not provided, but possible; s: t: transactions basis

¹ National account classifications are published with one-year delay; publication is planned to be suppressed.

 2 The Appropriation Bill contains a breakdown according to administrative function, however, this classification often does not identify the department acually in charge.

³ A breakdown is provided into 12 functions, leaving, however, a large amount of unallocated items.

⁴ Some information is provided.

Country	Procedures
Belgium	all but a few of approx. 100 special funds are now in the budget. Off-budget operations exist but are not published
Denmark	There are 17 special funds whose assets are used for earmarked purposes. Their expenditures are included under the government spending limit.
France	special funds for temporary operations are included in the budget.
Germany	Sondervermögen des Bundes are included with their overall revenue or expenditure implications.
Greece	Budgets of independent government bodies are annexed to budget and voted by Parliament. They belong to specialzed development or research oriented bodies and services relieved from rigidities of Public Accounting and subject to special financial regulations. If for any reason administration of revenue or expenditure is impossible on the basis of the budget, it can be made on the basis of a specific law for each time.
Ireland	Occasionally established funds to enable State to administer monies on behalf of private citizens or bodies (for example, if most funds come from sources other than general fund). plus: Social Insurance Fund, Post Office Bank, Sinking Fund and Lotterie Fund
Italy	Special funds exist for pending legislation with global expenditure estimates. Off-budget funds exist for certain ministries.
Luxembg.	n.a.
Netherlands	Off-budget operations exist at 0.4% of NNI. They are included in deficit data. Other special funds are included in budget.
Spain	
Portugal	Numerous off-budget operations existed until 1990 and were associated with fiscal indiscipline. There is a large number of autonomous funds and departments whose budgets are presented separately to Parliament, or whose budgets are only subject to government authorization.
U.K.	Little use of special funds, no use of off-budget operations.

Table A5: Treatment of Special Funds

F

Table A6: Voting Procedures

Country	Negotia Governm	Negotiations within Government			Parlia- Amendments ment can			joint vote on all G	global vote on budget	time limit	prov. if not met	
	agenda set by	type	budget negoti- ations	р г р	a m e n d	are ltd.	are off- setting	can cause fall of govern- ment				
Belgium	MB, MF	ь	С	n	у	n	n	У	У	A	у	12s
Denmark	с	s,G	bilat.	n	У	n	У	У	n	Α	У	12s
France	PM	b	bilat.	n	У	У	У	Уı	\mathbf{n}^2	В	У	TE'
Germany	MF	b	bilat.	у	у	n	n	У	n	Α	у	i
Greece	MF	G	с	n	У	n	n	n	У	A	У	ο
Ireland	с	D	с	n	У	y4	n	У	n	n	n	P
Italy	MF	b,s	multi- lat.	n	у	У	n	n	n	A ^s	у	max. 4/12
Luxembg.				У	у	Уé	n		У	Α	У	12s
Nether- lands	с	b,s	bilat.	n	У	n	р	У	n'	B*	n	4/12
Portugal	с	b,s	bilat.	n	У	n	n		n	A	У	0
Spain	с	b	C ,	n	У	У	- y ⁹	n	У	A	n	0
U.K.	MF 10	b	bilat.	n	у	y ¹¹	n	У	n	B ¹²	У	TE

Note: A: after general debate; b: general budget guidelines; B: before general debate; bilat: bilateral negotiations between MF and resort ministries; C: Cabinet; O: previous budget continued; PM: Prime Minister; s: specific budget targets determined; TE: temporary budget adopted.

¹ MPs may propose to reduce a receipt accompanied with an increase in another receipt, but not an increase in spending accompanied with a decrease in another spending item.

² Global vote on existing entitlements, vote chapter by chapter on new authorizations.

³ By decree of government.

* Parliament can only amend tax provisions, while expenditure proposals can only be refused or approved.

⁵ Parliament first votes on the Finance Bill which sets the overall ceiling for government borrowing. However, this ceiling being only in commitment terms, it is not a binding constraint.

⁶ Legally unrestricted, but very limited in practice.

⁷ Expenditures are voted by chapters, revenues separately.

⁸ Before discussing the individual chapters, there is a political discussion on the global content of the budget.

⁹ Unless authorized otherwise by government.

 10 MF submits proposal to be voted by C.

" Amendments to the Finance Bill may normally only reduce, but not increase taxes. See also fn. 11.

¹² The House of Commons first votes on the Budget resolutions which cannot be amended. The resolutions determine what goes into the Finance Bill and how that Bill might be amended. The result is that Government cannot be confronted with a budget completely different from its proposal and, provided it has a majority in the Commons, it can limit Parliament's amendments to what it deems acceptable.
Table A7: Budget Monitoring Rules

Country	Expenditure Control by	Finance Ministry records spending	Role of Parliament	Final Report Due After
Belgium	MF	monthly	informed Oct., March of t+1	up to five years
Denmark	Ministries	yes	informed quarterly	12 months
France	MF	monthly	informed monthly, debate in December	March t+1
Germany	Ministries and MF	monthly	no role ¹ , informed quarterly	4 months
Greece	MF	monthly	informed in November or at request	3 months
Ireland	MF	monthly	informed quarterly	12 months
Italy	M of Budget, MF	monthly	informed quarterly about D	6 months
Luxembg.	M of Budget	quar- terly	no role	
Nether- Lands	MF	monthly	informed quarterly	9 months
Portugal	Auditors only for State sector	monthly with delay	no role, regularly informed	9 months
Spain	MF and M of Economy	monthly	quarterly	
U.K.	MF	monthly	no role, bi-annually informed	3 months

¹ Parliamentary Committee (Haushaltsausschuss) monitores execution.

Country	MF can block expenditure	Cash Limits on Departments	Disbur- sement approval ¹	Transfers between Chapters	Budget changes authorized by	Carry-over to next year
Belgium			yes	within Depts ²	new law (March)	yes ³
Denmark		yes		yes⁴	new law	yes ^s
France	ଙ	yes	yes	lim, by decree	new law'	limited
Germany	yes	yes	yes	within Depts.*	MF°	possible ¹⁰
Greece	yes	yes		approved by MF	MF ¹¹	no
Ireland				within 'Votes' approved by MF	new law	no
Italy				yes	new law ¹²	yes
Luxembg.	yes			possible		
Nether- lands			rarely	yes ^{ıs}	Govt. ¹⁴	limited
Portugal		yes	often	within Depts. approv. by Parl.	new law	limited ¹⁵
Spain				limited	new law	limited
U.K.		yes		approved by MF ¹⁶	new law	limited ¹⁷

Table A8: Budget Flexibility

¹ By authority other than executive of ressort ministries, e.g. MF or financial controller.

² Transfers between Departments require rare political consensus.

³ Undifferentiated appropriations carried over by Royal Decree only; differentiated appropriations carried over automatically.

⁴ Only with approval by Finance Committee of Parliament.

⁵ But of little relevance in practice.

⁶ Government can block expenditure by decree.

⁷ In special cases, additional expenditure can be authorized by government decree which must be endorsed in the next budget law.

* Between departments, consent of MF is required.

* Upon initiative from government parties.

¹⁰ Requires consent of Parliament.

" MF can seek budget amendment during the discussion of the law closing the account or authorize spending over ceiling.

¹² To be submitted on June 30 or before October 31.

¹³ Cannot be used to finance new expenditure.

⁴ Additional spending must be compensated within the same chapter, save fore exceptions with approval by Cabinet. In such cases, general compensation is sought or emergency spending authorized.

¹⁵May be authorized for investment plans and autonomous or social security funds by the Budget Law.

" With approval by Parliament only for transfers between 'votes'.

" Five percent of capital expenditures and defense expenditures.

Index Construction for the Empirical Analysis

Item 1. Structure of negotiations within government

i

- a) general constraint: none (0), B/Y (1), B/Y and D/Y (2), G/Y or Golden Rule (3), G/Y, D/Y (4).
- b) agenda setting for budget negotiations: MF or cabinet collects bids from spending ministers (0); MF or cabinet collects bids subject to preagreed guidelines (1), cabinet decides on budget norms first (2), MF proposes budget norms to be voted on by cabinet (3), MF or MP determines budget parameters to be observed by spending ministers (4).
- c) scope of budget norms in the setting of agenda: expenditure or deficit (0), 'specific' (1.33), 'broad' and 'specific' (2.66), 'broad' (4).
- d) structure of negotiations: all cabinet members involved together (0), multilateral (2), bilateral between spending ministers and MF (4).

	General constraint	agenda setting	scope of budget norms	structure of negotiations
Belgium	0	1	1.33	0
Denmark	4	3	1.33	4
France	4	4	4	4
Germany	3	1	4	4
Greece	0	1	0	0
Ireland	2	1	0	0
Italy	2	1	2.66	2
Luxembourg	3	-	-	-
Netherlands	1	3	2.66	4
Portugal	1	2	2.66	4
Spain	0	2	4	0
U.K.	4	3	4	4

Item 2. Structure of parliamentary process

a) amendments: unlimited (0), limited (4)

b) required to be offsetting: no (0), yes (4)

- c) can cause fall of government: no (0), yes (4)
- d) all expenditures passed in one vote: yes (0), mixed (2), votes are chapter by chapter (4)
- e) global vote on total budget size: final only (0), initial (4)

	amendments limited	amendments offsetting	amendments cause fall	one vote on expenditure	global vote
Belgium	0	0	4	0	0
Denmark	0	4	4	4	0
France	4	4	4	2	4
Germany	0	0	4	0	0
Greece	0	0	0	0	0
Ireland	4	0	4	0	0
Italy	- 4	0	0	2	0
Luxembourg	4	0	-	0	0
Netherlands	4	0	4	4	4
Portugal	0	0	-	0	1
Spain	4	0	0	0	0
U.K.	4	0	4	4	4

Item 3. Informativeness of the budget draft

- a) special funds included:
 no (0), some (1), most (2), yes, but annexed to budget draft (3), yes
 (4).
- b) budget submitted in one document: no (0), recently yes (2), yes (4).
- c) assessment of budget transparency by respondents: hardly transparent (0), not fully transparent (2), fully transparent (4)
- d) link to national accounts: not provided (0), possible (1.33), provided in separate documents (2.66), direct link provided (4)
- e) government loans to non-government entities included in budget draft:
 no (0), reported in separate document (2), yes (4)

	special funds	one document	transparency	national accounts	government loans
Belgium	2	2	2	0	4
Denmark	2	4	2	1.33	4
France	4	4	4	2.66	-
Germany	3	4	4	4	2
Greece	3	0	4	1.33	2
Ireland	1	0	2	0	2
Italy	1	0	0	0	4
Luxembourg	-	4	2	1.33	-
Netherlands	4	4	2	4	4
Portugal	0	4	2	1.33	0
Spain	3	4	2	-	-
U.K.	4	0	4	4	4

- Item 4. Flexibility of Budget Execution
- a) MF can block expenditures: no (0), yes (4)
- b) spending ministries subject to cash limits: no (0), yes (4)
- c) disbursement approval required from MF or controller: no (0), yes (4)
- d) transfers of expenditures between chapters: unrestricted (0), limited (0.8), require consent of MF (1.6), require consent of parliament (2.4), only within departments possible (4), only within departments and with consent of MF (5).
- e) changes in budget law during execution: at discretion of government (0), by new law which is regularly submitted during fiscal year (1), at discretion of MF (2), require consent of MF and parliament (3), only by new budgetary law to be passed under the same regulations as the ordinary budget (4).
- f) carry-over of unused funds to next year: unrestricted (0), limited (1.33), limited and requires authorization by MF or parliament (2.66), not possible (4)

	MF can block	cash limits	disburse- ment approval	transfers	budget changes	carry- over
Belgium	0	0	4	3.2	4	0
Denmark	0	4	0	2.4	4	0
France	4	4	4	3.2	4	1
Germany	4	4	4	1.6	3	2
Greece	4	4	0	1.6	2	3
Ireland	0	0	0	4	4	3
Italy	0	0	0	0	1	0
Luxembourg	4	0	0	0	-	-
Netherlands	0	0	4	0	0	1
Portugal	0	4	4	0	4	2
Spain	0	0	0	0.8	4	1
U.K.	0	4	0	2.4	4	1

Item 5. Longterm Planning Constraint

a) multiannual target: none (0), G or T (2), total budget size (4)

b) planning horizon (years): two (1) three (2), four (3), five (4)

c) nature:

ad hoc forecast (1), fixed forecast (2), updated forecast, but not based on consistent macromodel (3), updated on basis of consistent macromodel (4)

d) degree of commitment:

internal orientation (1), indicative (2), weak political (3), strong political (4)

	target	horizon	nature	commitment
Belgium	0	0	0	0
Denmark	2	2	2	2
France	0	1	1	1
Germany	4	3	4	3
Greece	0	2	1	2
Ireland	4	4	1	3
Italy	4	3	1	3
Luxembourg	0	0	0	0
Netherlands	4	4	2	4
Portugal	0	3	1	2
Spain	0	4	1	1
U.K.	2	4	4	3

Definition of Indices:

1. Structural index:

SI1 = sum of row entries of itens 1. - 4.

SI2 = sum of row entries of items 1., 2. and 4.

SI3 = sum of row entries of items 1. and 2.

2. Index of longterm planning constraint:

CON1 - sum of row entries of items 5., 3., plus ammendment index plus flex
CON2 - sum of row entries of item 5. plus ammendment index plus flex
CON3 - sum of row entries of item 5 plus flex

where: amendment index is the sum of the first three row entries of item 2. and flex is the sum of the first, second, fourth and last rows of item 4.

Economic Papers

The following papers have been issued. Copies may be obtained by applying to the address mentioned on the inside front cover.

- No. 1 EEC-DG II inflationary expectations. Survey based inflationary expectations for the EEC countries, by F. Papadia and V. Basano (May 1981).
- No. 3 A review of the informal economy in the European Community, by Adrian Smith (July 1981).
- No. 4 Problems of interdependence in a multipolar world, by Tommaso Padoa-Schioppa (August 1981).
- No. 5 European Dimensions in the Adjustment Problems, by Michael Emerson (August 1981).
- No. 6 The bilateral trade linkages of the Eurolink Model : An analysis of foreign trade and competitiveness, by P. Ranuzzi (January 1982).
- No. 7 United Kingdom, Medium term economic trends and problems, by D. Adams, S. Gillespie, M. Green and H. Wortmann (February 1982).
- No. 8 Où en est la théorie macroéconomique, par E. Malinvaud (juin 1982).
- No. 9 Marginal Employment Subsidies : An Effective Policy to Generate Employment, by Carl Chiarella and Alfred Steinherr (November 1982).
- No. 10 The Great Depression : A Repeat in the 1980s ?, by Alfred Steinherr (November 1982).
- No. 11 Evolution et problèmes structurels de l'économie néerlandaise, par D.C. Breedveld, C. Depoortere, A. Finetti, Dr. J.M.G. Pieters et C. Vanbelle (mars 1983).
- No. 12 Macroeconomic prospects and policies for the European Community, by Giorgio Basevi, Olivier Blanchard, Willem Buiter, Rudiger Dornbusch, and Richard Layard (April 1983).
- No. 13 The supply of output equations in the EC-countries and the use of the survey-based inflationary expectations, by Paul De Grauwe and Mustapha Nabli (May 1983).
- No. 14 Structural trends of financial systems and capital accumulation : France, Germany, Italy, by G. Nardozzi (May 1983).
- No. 15 Monetary assets and inflation induced distorsions of the national accounts conceptual issues and correction of sectoral income flows in 5 EEC countries, by Alex Cukierman and Jorgen Mortensen (May 1983).
- No. 16 Federal Republic of Germany. Medium-term economic trends and problems, by F. Allgayer, S. Gillespie, M. Green and H. Wortmann (June 1983).
- No. 17 The employment miracle in the US and stagnation employment in the EC, by M. Wegner (July 1983).
- No. 18 Productive Performance in West German Manufacturing Industry 1970-1980; A Farrell Frontier Characterisation, by D. Todd (August 1983).
- No. 19 Central-Bank Policy and the Financing of Government Budget Deficits : A Cross-Country Comparison, by G. Demopoulos, G. Katsimbris and S. Miller (September 1983).
- No. 20 Monetary assets and inflation induced distortions of the national accounts. The case of Belgium, by Ken Lennan (October 1983).
- No. 21 Actifs financiers et distorsions des flux sectoriels dues à l'inflation : le cas de la France, par J.-P. Baché (octobre 1983).
- No. 22 Approche pragmatique pour une politique de plein emploi : les subventions à la création d'emplois, par A. Steinherr et B. Van Haeperen (octobre 1983).
- No. 23 Income Distribution and Employment in the European Communities 1960-1982, by A. Steinherr (December 1983).
- No. 24 U.S. Deficits, the dollar and Europe, by O. Blanchard and R. Dornbusch (December 1983).
- No. 25 Monetary Assets and inflation induced distortions of the national accounts. The case of the Federal Republic of Germany, by H. Wittelsberger (January 1984).

- No. 26 Actifs financiers et distorsions des flux sectoriels dues à l'inflation : le cas de l'Italie, par A. Reati (janvier 1984).
- No. 27 Evolution et problèmes structurels de l'économie italienne, par Q. Ciardelli, F. Colasanti et X. Lannes (janvier 1984).
- No. 28 International Co-operation in Macro-economic Policies, by J.E. Meade (February 1984).
- No. 29 The Growth of Public Expenditure in the EEC Countries 1960-1981 : Some Reflections, by Douglas Todd (December 1983).
- No. 30 The integration of EEC qualitative consumer survey results in econometric modelling : an application to the consumption function, by Peter Praet (February 1984).
- No. 31 Report of the CEPS Macroeconomic Policy Group. EUROPE : The case for unsustainable growth, by R. Layard, G. Basevi, O. Blanchard, W. Buiter and R. Dornbusch (April 1984).
- No. 32 Total Factor Productivity Growth and the Productivity Slowdown in the West German Industrial Sector, 1970-1981, by Douglas Todd (April 1984).
- No. 33 An analytical Formulation and Evaluation of the Existing Structure of Legal Reserve Requirements of the Greek Economy : An Uncommon Case, by G. Demopoulos (June 1984).
- No. 34 Factor Productivity Growth in Four EEC Countries, 1960-1981, by Douglas Todd (October 1984).
- No. 35 Rate of profit, business cycles and capital accumulation in U.K. industry, 1959-1981, by Angelo Reati (November 1984).
- No. 36 Report of the CEPS Macroeconomic Policy Group. Employment and Growth in Europe : A Two-Handed Approach by P. Blanchard, R. Dornbusch, J. Drèze, H. Giersch, R. Layard and M. Monti (June 1985).
- No. 37 Schemas for the construction of an "auxiliary econometric model" for the social security system, by A. Coppini and G. Laina (June 1985).
- No. 38 Seasonal and Cyclical Variations in Relationship among Expectations, Plans and Realizations in Business Test Surveys, by H. König and M. Nerlove (July 1985).
- No. 39 Analysis of the stabilisation mechanisms of macroeconomic models : a comparison of the Eurolink models by A. Bucher and V. Rossi (July 1985).
- No. 40 Rate of profit, business cycles and capital accumulation in West German industry, 1960-1981, by A. Reati (July 1985).
- No. 41 Inflation induced redistributions via monetary assets in five European countries : 1974-1982, by A. Cukierman, K. Lennan and F. Papadia (September 1985).
- No. 42 Work Sharing : Why ? How ? How not ..., by Jacques H. Drèze (December 1985).
- No. 43 Toward Understanding Major Fluctuations of the Dollar by P. Armington (January 1986).
- No. 44 Predictive value of firms' manpower expectations and policy implications, by G. Nerb (March 1986).
- No. 45 Le taux de profit et ses composantes dans l'industrie française de 1959 à 1981, par Angelo Reati (Mars 1986).
- No. 46 Forecasting aggregate demand components with opinions surveys in the four main EC-Countries Experience with the BUSY model, by M. Biart and P. Praet (May 1986).
- No. 47 Report of CEPS Macroeconomic Policy Group : Reducing Unemployment in Europe : The Role of Capital Formation, by F. Modigliani, M. Monti, J. Drèze, H. Giersch and R. Layard (July 1986).
- No. 48 Evolution et problèmes structurels de l'économie française, par X. Lannes, B. Philippe et P. Lenain (août 1986).
- No. 49 Long run implications of the increase in taxation and public debt for employment and economic growth in Europe by G. Tullio (August 1986).
- No. 50 Consumers Expectations and Aggregate Personal Savings by Daniel Weiserbs and Peter Simmons (November 1986).
- No. 51 Do after tax interest affect private consumption and savings ? Empirical evidence for 8 industrial countries : 1970-1983 by G. Tullio and Fr. Contesso (December 1986).
- No. 52 Validity and limits of applied exchange rate models : a brief survey of some recent contributions by G. Tullio (December 1986).

- No. 53 Monetary and Exchange Rate Policies for International Financial Stability : a Proposal by Ronald I. McKinnon (November 1986).
- No. 54 Internal and External Liberalisation for Faster Growth by Herbert Giersch (February 1987).
- No. 55 Regulation or Deregulation of the Labour Market : Policy Regimes for the Recruitment and Dismissal of Employees in the Industrialised Countries by Michael Emerson (June 1987).
- No. 56 Causes of the development of the private ECU and the behaviour of its interest rates : October 1982 - September 1985 by G. Tullio and Fr. Contesso (July 1987).
- No. 57 Capital/Labour substitution and its impact on employment by Fabienne Ilzkovitz (September 1987).
- No. 58 The Determinants of the German Official Discount Rate and of Liquidity Ratios during the classical goldstandard : 1876-1913 by Andrea Sommariva and Giuseppe Tullio (September 1987).
- No. 59 Profitability, real interest rates and fiscal crowding out in the OECD area 1960-1985 (An examination of the crowding out hypothesis within a portfolio model) by Jorgen Mortensen (October 1987).
- No. 60 The two-handed growth strategy for Europe : Autonomy through flexible cooperation by J. Drèze, Ch. Wyplosz, Ch. Bean, Fr. Giavazzi and H. Giersch (October 1987).
- No. 61 Collusive Behaviour, R & D, and European Policy by Alexis Jacquemin (November 1987).
- No. 62 Inflation adjusted government budget deficits and their impact on the business cycle : empirical evidence for 8 industrial countries by G. Tullio (November 1987).
- No. 63 Monetary Policy Coordination Within the EMS : Is There a Rule ? by M. Russo and G. Tullio (April 1988).
- No. 64 Le Découplage de la Finance et de l'Economie Contribution à l'Evaluation des Enjeux Européens dans la Révolution du Système Financier International par J.-Y. Haberer (Mai 1988).
- No. 65 The completion of the internal market : results of macroeconomic model simulations by M. Catinat, E. Donni and A. Italianer (September 1988).
- No. 66 Europe after the crash : economic policy in an era of adjustment by Charles Bean (September 1988).
- No. 67 A Survey of the Economies of Scale by Cliff Pratten (October 1988).
- No. 68 Economies of Scale and Intra-Community trade by Joachim Schwalbach (October 1988).
- No. 69 Economies of Scale and the Integration of the European Economy : the Case of Italy by Rodolfo Helg and Pippo Ranci (October 1988).
- No. 70 The Costs of Non-Europe An assessment based on a formal Model of Imperfect Competition and Economies of Scale by A. Smith and A. Venables (October 1988).
- No. 71 Competition and Innovation by P.A. Geroski (October 1988).
- No. 72 Commerce Intra-Branche Performances des firmes et analyse des échanges commerciaux dans la Communauté européenne par le Centre d'Etudes Prospectives et d'Informations Internationales de Paris (octobre 1988).
- No. 73 Partial Equilibrium Calculations of the Impact of Internal Market Barriers in the European Community by Richard Cawley and Michael Davenport (October 1988).
- No. 74 The exchange-rate question in Europe by Francesco Giavazzi (January 1989).
- No. 75 The QUEST model (Version 1988) by Peter Bekx, Anne Bucher, Alexander Italianer, Matthias Mors (March 1989).
- No. 76 Europe's Prospects for the 1990s by Herbert Giersch (May 1989).
- No. 77 1992, Hype or Hope : A review by Alexander Italianer (February 1990).
- No. 78 European labour markets : a long run view (CEPS Macroeconomic Policy Group 1989 Annual Report) by J.-P. Danthine, Ch. Bean, P. Bernholz and E. Malinvaud (February 1990).
- No. 79 Country Studies The United Kingdom by Tassos Belessiotis and Ralph Wilkinson (July 1990).
- No. 80 See "Länderstudien" No. 1
- No. 81 **Country Studies** The Netherlands by Filip Keereman, Françoise Moreau and Cyriel Vanbelle (July 1990).

- No. 82 Country Studies Belgium by Johan Baras, Filip Keereman and Françoise Moreau (July 1990).
- No. 83 Completion of the internal market : An application of Public Choice Theory by Manfred Teutemann (August 1990).
- No. 84 Monetary and Fiscal Rules for Public Debt Sustainability by Marco Buti (September 1990).
- No. 85 Are we at the beginning of a new long term expansion induced by technological change?, by Angelo Reati (August 1991).
- No. 86 Labour Mobility, Fiscal Solidarity and the Exchange Rate Regime: a Parable of European Union and Cohesion, by Jorge Braga de Macedo (October 1991).
- No. 87 The Economics of Policies to Stabilize or Reduce Greenhouse Gas Emissions: the Case of CO2, by Mathias Mors (October 1991).
- No. 88 The Adequacy and Allocation of World Savings, by Javier Santillán (December 1991).
- No. 89 Microeconomics of Saving, by Barbara Kauffmann (December 1991).
- No. 90 Exchange Rate Policy for Eastern Europe and a Peg to the ECU, by Michael Davenport (March 1992).
- NO. 91 The German Economy after Unification: Domestic and European Aspects, by Jürgen Kröger and Manfred Teutemann (April 1992).
- No. 92 Lessons from Stabilisation Programmes of Central and Eastern European Countries, 1989-91, by Domenico Mario Nuti (May 1992).
- No. 93 Post-Soviet Issues: Stabilisation, Trade and Money, by D. Mario Nuti and Jean Pisani-Ferry (May 1992).
- No. 94 Regional Integration in Europe, by André Sapir (September 1992).
- No. 95 Hungary : Towards a Market Economy (October 1992).
- No. 96 Budgeting Procedures and Fiscal Performance in the European Communities, by Jürgen von Hagen (October 1992).

Country Studies*

See also Economic Papers No.79 (The United Kingdom), No.81 (The Netherlands) and No.82 (Belgium).

- No. 1 The Federal Republic of Germany (September 1990)
- No. 2 Portugal (February 1991)
- No. 3 United Kingdom (March 1991)
- No. 4 Denmark (April 1991)
- No. 5 France (août 1991)
- No. 6 Ireland (September 1991)
- No. 7 Spain (March 1992)
- No. 8 Netherlands (June 1992)
- No. 9 Greece (July 1992)

^{*} Country studies are the result of internal analysis of the economic situation of the respective member country; they are made on the responsibility of the Directorate-General for Economic and Financial Affairs of the Commission of the European Communities.